

VUE SOFTWARE MANUAL

Using VUE software to view and organize data | version 2.7



innovasea.com/fish-tracking

DOC-4399-32

SUMMARY OF MANUAL

This manual is intended to provide the information required to use **VUE software**. Innovasea recommends that you fully read the manual before using VUE.

Section 1: Introduction

Getting to know what VUE software can do for you

Section 2: Getting VUE Started How to get VUE ready to begin a study (i.e. before you deploy your receivers)

- Section 3: Data Analysis the basics Getting your data ready for analysis
- Section 4: Viewing Data

How to view your data in VUE, including using the filter feature

Section 5: Managing Data

How to import and export data, and how to view the millisecond timestamps

Section 6: Viewing Events

To view event information

Section 7: Additional Information Helpful information, like how to install the software and set

Section 8: Appendices

Information pertaining to using VUE software with Innovasea receivers is located in the user manual for that receiver. For example, to learn how to communicate with a VR2W, go to the VR2W User Manual.

TABLE OF CONTENTS

1	Introd	uction	.7
	1.1 Ove	rview of VUE Software	7
	1.1.1	What can VUE do?	7
	1.1.2	What's a VRL file?	8
	1.2 The	order of things	9
	1.3 Set	ing the correct UTC time	9
	1.4 Soft	ware Tabs - an overview	.10
	1.4.1	Detections Tab	.11
	1.4.2	Events Tab	.12
	1.5 Get	ting Help	.12
2	Gettin	g VUE Started	13
	2.1 Ope	en or Create a Database	.13
2	2.2 Sav	ing a database copy	.14
	2.3 Set	Up Tags	.15
	2.3.1	Add Information for a New Sensor Tag	.16
	2.3.2	Edit Information for an Existing Sensor Tag	.17
	2.3.3	Predation Sensor Tags	.18
	2.3.3	Predation Sensor Tags	.18
,	2.3.3.1	Interpreting Predation Sensor Data in Vue	.18
4			.21
	2.4.1	Add a New Station	.22
	2.7.2		.20
3	Data A	nalysis - The basics	25
3	3.1 Tim	e Correction	.25
3	3.2 Fals	se Detection Analysis Tool	.28
	3.2.1	Using the FDA tool	.28
	3.2.2	Reviewing "Questionable" Data	.30
	3.2.3	Changing FDA Interval times	.30
	3.3 Res	idency Search	.31
4	Viewir	ng Data	34
2	4.1 Viev	ving Detections	.34
	4.1.1	Detections Ribbon	.35
	4.1.2	Detections in the Selection Tree	.35
	4.1.3	Viewing All Detections.	.35
	4.1.4	Viewing Detections by Receiver	.36
	4.1.5 / 1.6	Viewing Transmitter Detections	.30
	4.1.0 4.1.6.1	Transmitter Name	.37
	4.1.7	Viewing by Station	.37
		o ,	

4.2 Plotting Data	42
4.2.1 Visualizing Predation Events	41
4.3 Filtering Data	
4.3.1 Creating Detection Filters	
4.3.2 An Example of a Common Filter	
4.3.3 Filter Options	
4.3.5 Exporting Detection Filters to Another Database	
5 Managing Data	
5.1 Importing Data	47
5.2 Editing a VRL file	50
5.2.1 Correcting or Adding a Station Name to a VRL file	51
5.3 Managing Duplicate Tags in VUE	
5.3.1 What is a duplicate tag?	
5.3.2 Verifying a transmitter and its data	
5.4 Removing files from VUE	
5.5 Removing Transmitters from VUE	
5.6 Viewing Millisecond Timestamps	
5.6.1 Viewing millisecond timestamps in MS Excel	
5.7 Exporting Data	
5.8.1 What is Matadata?	60
5.8.1 What is Metadata? 5.8.2 Transferring Metadata Between Databases	60
 5.8.1 What is Metadata? 5.8.2 Transferring Metadata Between Databases 	
 5.8.1 What is Metadata?	60
 5.8.1 What is Metadata?	60
 5.8.1 What is Metadata?	60
 5.8.1 What is Metadata?	60 60 62 62 62 63 63 63 63 63 63 63 63 64 64 64 64 64 64 64 64 64 65 66 67
 5.8.1 What is Metadata?	60
 5.8.1 What is Metadata?	60
5.8.1 What is Metadata? 5.8.2 Transferring Metadata Between Databases 6 Viewing Events 6.1 Events Tab 6.1.1 Grouping Events by Type 6.2 Events Ribbon 6.3 Events n the Selection Tree 6.4 Events Common to all Receiver Types 6.4.1 Reset Event 6.4.2 Initialization Event 6.4.3 Memory Capacity Event 6.4.4 Receiver-Specific Data Summaries 6.5 Exporting Events 7 Additional Information 7.1 Installing the VUE software 7.2 Connecting with Bluetooth 7.2.1 Using INTERNAL Bluetooth Adapter	60 60 62 62 62 63 63 63 63 63 63 63 63 63 63 63 63 63
 5.8.1 What is Metadata?	60
 5.8.1 What is Metadata?	60 60 62 62 63 63 63 63 63 63 63 63 63 63 63 64 64 64 64 64 64 64 64 64 64 64 64 66 66

0		entact Information	
Q	Anno	ndicos	72
	7.4.6	Other VUE Settings	71
	7.4.5	FDA Tool Settings	71
	7.4.4	Devices	70
	7.4.3	Export Settings	70
	7.4.2	Offload Settings	69

 \bigcirc (2020) Innovasea. All rights reserved. The information contained herein includes information which is confidential and proprietary to Innovasea and may not be used or disclosed without prior written consent of Innovasea.

The *Bluetooth*® word mark and logos are owned by the Bluetooth SIG, Inc. and any use of such marks by Innovasea is under license. Other trademarks and trade names are those of their respective owners.

For the latest versions of user manuals and software, visit support.fishtracking.innovasea.com

I. INTRODUCTION

1.1 Overview of VUE Software

The VUE software has been developed to aid researchers in gathering, viewing, and analyzing acoustic detection data from Innovasea equipment. VUE employs a central database to allow the collection and viewing of data from multiple Innovasea receivers of various types. The VUE software is compatible with VR2W, VR2Tx, VR2AR, VR2C, and VR4-UWM receivers as well as the Ascent acoustic release. Legacy VR2 and VR3-UWM data files can also be imported into VUE databases.

VUE is used to perform such tasks as retrieving receiver data files, setting the receiver's clock, configuring the receiver's built-in transmitter (if applicable), and clearing the receiver's memory. Information on using VUE for receiver-specific communication is located in the individual receiver user manuals.

VUE can be used to create and manage one or more central databases combining data files from multiple receivers and multiple types of receivers. Having a central database allows easy comparison between detections from multiple studies, across multiple years, and from various locations. You may choose to create smaller, temporary databases on laptops used for collecting data in the field.

VUE databases combine detection data from multiple receivers.

A larger, more central, database containing all receiver data files can then be created on a lab computer or network server.

Always use the latest version of VUE on your *entire suite* of PC's and laptops. This will ensure access to the latest software features and improvements and compatibility with newer and older Innovasea receivers.

1.1.1 What can VUE do?

VUE can...

- 1. Detect and report a misconfigured computer clock (see section 1.3 Setting correct UTC time)
- 2. Setup receivers to record data (details found in related receiver manual)
- 3. Collect data from receivers and create VRL files (details found in related receiver manual)
- 4. Adjust data for clock-drift (see section 3.1)
- 5. Filter data within the database (see section 4.3)
- 6. Deal with duplicate tags (see section 5.3)
- 7. Identify potential false detections for review (FDA tool, see section 3.2)
- 8. View data in graphical form (see section 4.2)
- 9. Export data (see section 5.7)
- 10. Update receiver firmware (details found in related receiver manual)



1.1.2 What's a VRL file?

A VRL file is created each time data is offloaded from a receiver using the VUE software¹. VRL files are digitally encrypted/signed binary files that cannot be altered and provide a permanent record of the data logged by

the receiver. A copy of these log files should be stored in a safe location as a backup of the collected data. If you ever need to restore your VUE database, you can import the original data from your VRL files. We suggest uploading your receiver log files to your Fathom Central account (fathomcentral.com) for secure backup and access to data management and visualization tools.



A VRL file contains receiver configuration information, diagnostic data, event records and the detection data. VRL files can be imported into any number of VUE databases and/or your Fathom Central account.

VRL files are named to allow you to identify the receiver to which they belong and the date that the data was offloaded. For example, the file VR2W_103047_20131130_1.VRL came from a VR2W receiver with the serial number 103047 and was offloaded on November 30, 2013. The "1" at the end indicates that it was the first offload of the day for that receiver using a particular PC.

VRL files with the RLD identifier in the filename (RLD data file, as in "VR2AR-RLD_545565_20141216_1.VRL") cannot be imported into VUE.

¹ VRL files are a different log file format than VDAT log files. VDAT log files are created when data are offloaded via the Fathom Mobile mobile app or the Fathom Connect desktop software. VDAT files can be converted to VRL file format for import into VUE using our online tool at fathomcentral.com.

1.2 The order of things...

A common question is "*What tasks do I perform and in what order am I supposed to perform them*?" The list below contains common tasks in a popular order.

- 1. Install VUE software (see section 7.1)
- 2. Set the correct UTC time (see section 1.3)
- 3. Setup VUE options (see section 7.4)
- 4. Create a database, if one doesn't yet exist (see section 2.1)
- 5. Setup sensor tags and stations (see sections 2.3 and 2.4)
- 6. With the receiver(s) (details found in related receiver manual):
 - a. Initialize the study
 - b. Test receiver(s) in-air and in water
 - c. Deploy receivers and tags and wait...
 - d. Offload data from receivers
- 7. Time correct data (see section 3.1)
- 8. Import data into database, if not imported during Time Correction (see section 5.1)
- 9. Identify any potential false detections and remove as required (see section 3.2)
- 10. View data (see section 4)
- 11. Export data, if desired (see section 5.7)

1.3 Setting the correct UTC time

Each time you open VUE while connected to the internet, VUE will automatically obtain the current correct UTC time from an internet time server. VUE will then calculate your computer's UTC time based on your selected clock settings (time zone and displayed time of day). If your computer's UTC time does not match the time obtained from the internet, VUE will issue a "PC Clock Configuration Warning".

This warning indicates that your computer's displayed time of day is not consistent with your selected time zone.

Your computer	's time settings do no	ot match internet time	
Please ensure that yo initializing or offload	our computer's time is correct	for your selected time zone before	
or ornood	Clock Time	Time Zone	
PC Local Time	2015-09-09 13:36:51	UTC-03:00 Atlantic Daylight Time	
Internet Time	2015-09-09 15:36:52	UTC-03:00 Atlantic Daylight Tin	
	Time discrepancy: 2h 0m	1s	

.

Innovasea **strongly recommends** correcting your computer's time settings *before* proceeding with your VUE session. The UTC time that VUE calculates from these settings is used to set your receiver clocks when starting a study and to calculate receiver clock drift when you time correct your VRL files.

Correct computer clock settings at both initialization and offload are critical to ensuring correct detection times in your data.

To correct your computer's clock settings, click "Adjust PC time settings"; the Windows Date and Time

Adjust your **time zone** to the desired setting (click **Change time zone...** button) and then make sure that the **displayed time of day is correct for the time zone that you have selected** (click **Change date and time...** button).

VUE's PC Clock Configuration Warning box (see previous) will update the Discrepancy field (time discrepancy between your computer's time settings and internet time, displayed in the time zone you that have selected) with each change you make to your computer's clock settings. It is recommended that you set your computer's clock to **within 2 minutes** of internet time.

📸 Date and Time	×
Date and Time Additional Clod	ks
	Date: Tuesday, August 18, 2015 Time: 1:13:56 AM Change date and time
Time zone	
(UTC-04:00) Atlantic Time	(Canada)
	Change time zone
Daylight Saving Time ends clock is set to go back 1 h	s on Sunday, November 01, 2015 at 2:00 AM. The bur at that time.
Got more time zone inform	action online
How do I set the clock and	I time zone?
	OK Cancel Apply

When satisfied that your clock settings are correct, click **Done**.

You may choose to ignore VUE's PC Clock Configuration Warning (by clicking **Done** without changing your clock settings), should you wish to correct your clock settings at a later time. This is, however, **not recommended**, and VUE will re-issue this warning should your clock settings remain incorrect and you attempt to **start** a new VUE session, **initialize** a receiver, or **offload** a receiver.

1.4 Software Tabs - an overview

VUE has a number of tabs allowing access to detections, events, and receivers depending if a database is open and/or a receiver is connected.

The File tab allows for databases to be created, opened, and closed, settings to be altered, and VRL files to be edited. The Home tab is used when connecting to a receiver and is explained in the receiver user manual. The Receiver tab is used, and only available, when VUE is communicating with a receiver. Multiple receivers can be in communication with VUE at once, resulting in more than one receiver tab open (see receiver manual for more information).

The tabs used during data manipulation are the Detections tab (see sections 1.4.1 and 4) and the Events tab (see sections 1.4.2 and 6). This VUE manual focuses on the functions of these two tabs. The Home and Receiver tabs are discussed in the receiver user manuals.



1.4.1 Detections Tab

The Detections tab contains all the detections in the open database and is only visible if a database is open. From here, we can graph detections (section 4.2), search data for animal residency (section 3.3), import (section 5.1) and export files (section 5.7), filter the data (section 4.3), add a new station (section 2.4.1) or transmitter record (section 2.3.1), and view data by receiver, transmitter, or station (section 4.1).



1.4.2 Events Tab

The Events tab contains a list of key receiver-related events in the currently open database as well as diagnostic information such as pings counts and logged noise and tilt information (where available). The Events tab is only visible if a database is open. From here, we can view events by receiver, import and export files, and delete individual log files. Event information is explained in more detail in section 6.



1.5 Getting Help

While VUE is open, you can access the user manual for VUE and for the receivers that are used with VUE by selecting the "Help" icon near the top right corner of the window (blue circle with question mark). When this icon is selected, a list of the available manuals will appear, and you can select the manual related to your receiver model. For example, if you have a question about communicating with the VR2W, then select the VR2W manual.

The manuals in VUE Help were current when the software was compiled. See <u>support.fishtracking.innovasea.com</u> for the latest version of the manuals.



2. GETTING VUE STARTED

This section of the VUE manual has the necessary information to get you started using VUE for your data management and analysis. After you've collected the data from your receivers as explained in the user manual for your receiver(s), the remaining sections of this manual will explain how to prepare, view, and export your data.

2.1 Open or Create a Database

To view data that has been offloaded from Innovasea receivers, first create a new database or open an existing database.

Begin by selecting the File menu button in the top left corner of the VUE software (circled at right). From the menu that appears, choose "New Database" to create a new database or choose "Open Database" to open an existing database (see below).



File menu button

File	
New Database	
Open Database	
As	
Close Database	
Tools .	
About VUE	
	Options 📲 Exit VUE

Regardless of your choice, the window entitled "Open or Create a new database" opens to allow you to select an existing database from the list shown or to enter the name of the new database in the "File name" box.



2.2 Saving a database copy

VUE allows you to save any currently open database under a different name (as a backup copy, for example).



Innovasea recommends making a copy of your VUE database prior to making significant edits, since database changes are irreversible.

To save an open database under a different name, select the File menu button in the top left corner of the VUE software. From the menu that appears, choose "Save Database As...". The "Save Database As..." window appears.

			·	-
)rganize 🔻 Nev	v folder			• 🔞
Favorites	•	Name	Date modified	Туре 🖍
		Dawn_Lake-2014 - Copy.vdb	8/26/2015 4:49 PM	VDB Fi
🔰 Libraries		Dawn_Lake-2014.vdb	8/19/2015 2:18 PM	VDB Fi ≡
Documents		Ross_Lagon-2012.vdb	8/27/2015 1:36 PM	VDB Fi
J Music	=	Ross_Lagon-2013.vdb	8/25/2015 4:30 PM	VDB Fi
Pictures				
📑 Videos				-
	-	•		F.
File name:	Ross_L	agon-2012 - Copy.vdb		•
Save as type:	Vemco	Database (*.vdb)		•

About VUE



In the File Name box, change the name of the file (example: [FileName]-copy.vdb), and click Save. By default, the database copy is saved in the same folder as the original file.

2.3 Set Up Tags

Detections of both coded pingers (tags without sensors) and coded sensor tags will be displayed in VUE upon import of VRL file(s) with no additional setup. Uncalibrated sensor data are reported as Analog to Digital Converter units, or ADC. For all sensor tags except predation tags (section 2.3.3), the slope and intercept of the sensor must be entered into VUE to calibrate and view the data in the applicable "real world" units.

Coded tags will be detected without being setup.

Slope and intercept information can be found in the Transmitter Specifications printout received with your tags. You may also access and download your tag specifications by logging in to your Customer Support Community (<u>support.fishtracking.innovasea.com</u>) account. If you prefer to import sensor tag calibration information from a file, please request a .VXM file from your Innovasea Sales or Support representative and import this file into VUE². Detailed instructions for importing sensor tag metadata from a .VXM file can be found at <u>support.fishtracking.innovasea.com</u>.

When entering sensor tag slope and intercept information manually, these values can be entered either before the tag is deployed (section 2.3.1) or after the tag has been detected and imported into the database (section 2.3.2).

² When importing sensor tag calibration information from a VXM file, be sure to import the VXM file *before* loading any detections of the specified sensor tags into your VUE database. Importing a VXM file after detections of the sensor tags already exist in the database will result in the creation of duplicate tags in the database

Tag calibration information is not stored in the receiver but instead is kept within the VUE database. Therefore, it is not necessary to enter tag calibration information into VUE before initializing and deploying a receiver. Tag calibration information can be exchanged between databases using the Export Metadata and Import Metadata features described in section 5.8.

2.3.1 Add Information for a New Sensor Tag

If a sensor tag detections are not yet present in a VUE database, use the steps below to prepare the database for sensor tag data to be displayed in the proper units (units selected in the *Options* window, section 7.4.1).

If you prefer to import sensor tag calibration information from a file, please request a .VXM file from your Innovasea Sales or Support representative and import this file into VUE. Detailed instructions for importing sensor tag metadata from a .VXM file can be found at support.fishtracking.innovasea.com.

STEP 1

Open the *Add Transmitter* window by either clicking on the "Add a new transmitter record" icon in the Detections ribbon (shown below) or by

right-clicking on the transmitter list and selecting "Add Transmitter" (shown at far right).



* All Detections	
🌾 Receivers (0)	Date
Transmitter Add Transmitter	
Imported Log Files (0) Totection Filters (0)	

STEP 2

Enter the information in the *Add Transmitter* window. This information - including whether your tags can transmit an error code - is provided in the Transmitter Specifications printout that was shipped with the tags. For more information on transmitter error codes, please search "Transmits error code" on <u>support.fishtracking.innovasea.com</u>.

Click "OK" when finished.



If you choose to name the tag, DO NOT use the same name for more than one tag. This causes ambiguity in the database.

Coding Type:	9007 - Coded Sensor 🗸 🗸
Frequency:	69.0 kHz
Tag ID:	11236
Tag S/N:	1234567
Tag Name:	(optional)
Sensor 1	
Type:	Temperature V
Slope:	0.1569
Intercept:	0
	✓ Transmits error code
	OK Cancel

TIP: To ease the task of adding sensor tag information to VUE, try importing the detection data collected during your in-air testing (instructions found in receiver manuals; test every sensor tag). Once the data are in VUE, follow the instructions in section 2.3.2 to edit the tag information to include the serial number and calibration information.

2.3.2 Edit Information for an Existing Sensor Tag

If sensor tag detections are present in a VUE database but sensor tag calibration information has not been setup, then the steps below will allow you to enter the necessary information for the data to be displayed in the proper units (units are selected in the *Options* window, section 7.4.1). As mentioned in the tip found in section 2.3.1, it is sometimes easier to enter a sensor's information after the tag has been detected by a receiver. In-air testing performed before the equipment is deployed, as described in the receiver manuals, not only ensures the equipment is functioning properly but can also be used to aid in entering sensor tag information.





If you choose to name the tag, DO NOT use the same name for more than one tag. This causes ambiguity in the database.

OK

Cancel

2.3.3 Predation Sensor Tags

When predation is detected by a predation sensor tag, it will transmit a sensor value that provides details on the predation status and the amount of time since the sensor was triggered³. The table below shows the meaning of unconverted raw sensor values (ADC units) for predation sensor tags.

Sensor Value (ADC Units)	Meaning
0	Sensor Fault
1	Untriggered
2	Triggering
3-254	Count of time since triggered
255	Maximum time counter reached

The precision of a predation sensor's time since trigger counter varies with the time since the sensor was triggered. As time since sensor trigger increases, time resolution decreases according to the table below.

Time Since Sensor Trigger [*] (Days)	Time Since Sensor Trigger [*] (Hours)	Sensor Value (ADC Units)	Precision
0 – 7 days	0 – 168 hours	3-170	±0.5 hour
7 – 14 days	170 – 334 hours	171-212	±2 hours
14 – 21 days	340 – 500 hours	213-233	±4 hours
21 – 35 days	512 – 832	234-254	±8 hours
>35 days	≥ 840	255	-

^{*}Please note that the delay between the predation event and the sensor trigger event is dependent on a number of external factors including water temperature and predator meal size.

To view time since sensor trigger in VUE, follow the steps described in Section 2.3.1 (*Add Information for a New Sensor Tag*) or Section 2.3.2 (*Edit Information for an Existing Sensor Tag*) and select "Predation 35d" from the Type dropdown list. Note that predation sensor tag calibration does *not* require input of Slope and Intercept information.

³ Please note that 180 kHz V5D predation tags purchased prior to April 2020 function differently than indicated above. Earlier V5D tags were not sensor tags; instead, they change ID upon triggering. They did not report time since trigger.

Tag S/N:		
Tag Name:		(optional)
Sensor 1		
Type:	Predation 35d V	
Slope:		
Intercept:		
	Transmits error code	

2.3.3.1 Interpreting Predation Sensor Data in VUE

The database shown below contains detections of a predation sensor tag. The "Data" column shows either sensor status (Sensor Fault, Untriggered, or Triggering), or, if triggered, count of time since triggered.

To interpret displayed count of time since triggered, refer to the table in Section 2.3.3 above. For example, for the detection highlighted in red below, the Data column indicates 178 h since the sensor was triggered. Referring to the table in Section 2.3.3, we see that at 178 h trigger time, sensor precision is ± 2 hours. Therefore, this sensor tag triggered between 176 and 180 hours before it was detected.

V Predation Sensor Demo.vdb							
File Home Detections	Events						
√ - ∕∕ - ∂	.	7	7.7		• •× •/	•} •})
Datadiana			Eillene	0	Chattioner	Transmitte	
All Detections	riles		riiters		Stations	Iransmitte	rs
All Detections All Detections	L	1				1	1
 Transmitters (2) 	Date	Time	Code Space	ID	Receiver	Station	Data
A69-1601-62011	2020 11 10	10.20.20	A69-9007	11236	VR2Tx-483190		Sensor Fault
s Interview Predation_01	2020 11 10		A69-9007	11236	VR2Tx-483190		Untriggered
Stations (0)	2020 11 10		A69-9007	11236	VR2Tx-483190		Triggering
> Imported Log Files (1)	2000	10.01.00	A69-9007	11236	VR2Tx-483190		0.5 h
T Detection Filters (0)	2020 11 10	10.01.00	A69-9007	11236	VR2Tx-483190		1.5 h
	2020 11 10	10.00.10	A69-9007	11236	VR2Tx-483190		2.5 h
	2000 11 10	1	A69-9007	11236	VR2Tx-483190		3.5 h
	2020 11 10		A69-9007	11236	VR2Tx-483190		170.0 h
	2020 11 10	10.00.07	A69-9007	11236	VR2Tx-483190		174.0 h
	2020 11 10		A69-9007	11236	VR2Tx-483190		178.0 h
	2000 10 10	10.04.00	A69-9007	11236	VR2Tx-483190		182.0 h
	2000-10-10		A69-9007	11236	VR2Tx-483190		186.0 h
	2000-11-10		A69-9007	11236	VR2Tx-483190		340.0 h
	2000 10 10		A69-9007	11236	VR2Tx-483190		348.0 h
	2020 11 10		A69-9007	11236	VR2Tx-483190		356.0 h
	2000 10 10	10.07.40	A69-9007	11236	VR2Tx-483190		364.0 h
	2020 11 10		A69-9007	11236	VR2Tx-483190		372.0 h
	2000.11.10	10.00.00	A69-9007	11236	VR2Tx-483190		380.0 h
	2020 11 10		A69-9007	11236	VR2Tx-483190		512.0 h
	2000 10 10	10.00.00	A69-9007	11236	VR2Tx-483190		528.0 h
	2000 10 10		A69-9007	11236	VR2Tx-483190		544.0 h
	2020 11 10	10.11.00	A69-9007	11236	VR2Tx-483190		560.0 h
	2000-00-00		A69-9007	11236	VR2Tx-483190		576.0 h

Note that when predation sensor detection data are exported from VUE to a CSV file, sensor precision information is included in the "Sensor Precision" column of the CSV file:

Date and Time (UTC)	Receiver	Transmitter	Transmitter Serial	Sensor Value	Sensor Unit	Transmitter Type	Sensor Precision
		A69-9007-11236	1234567	Sensor Fault	state	Predation 35d	
2020 11 10 10:30	VR2T. 100190	A69-9007-11236	1234567	Untriggered	state	Predation 35d	
	1027. 000100	A69-9007-11236	1234567	Triggering	state	Predation 35d	
2020 11 10 10:01	UR2T. 100190	A69-9007-11236	1234567	0.5	h	Predation 35d	0.5
3020 11 10 10:01	UR2T. 100190	A69-9007-11236	1234567	1.5	h	Predation 35d	0.5
20101 21 20 2000	10070-000200	A69-9007-11236	1234567	2.5	h	Predation 35d	0.5
2010 10 10 10100		A69-9007-11236	1234567	3.5	h	Predation 35d	0.5
3020 11 10 10100	1027. 000100	A69-9007-11236	1234567	170	h	Predation 35d	2
1000 10 10 10 10	MARTIN MALAN	A69-9007-11236	1234567	174	h	Predation 35d	2
ALC: 1 1 1		A69-9007-11236	1234567	178	h	Predation 35d	2
	Provide and the second	A69-9007-11236	1234567	182	h	Predation 35d	2
100 C C 000		A69-9007-11236	1234567	186	h	Predation 35d	2
ALC: 1 1 1		A69-9007-11236	1234567	340	h	Predation 35d	4

2.4 Set Up Stations

As receivers are portable units easily moved from one location to another, VUE has provided the option to associate a receiver with a station (location) for a particular deployment. This is setup when you configure the receiver (see *Start a Study* in the receiver manual). The station specifies a location, including latitude and longitude (if desired), which is assigned and recorded in the receiver's memory during initialization.

A station is used to identify the location of the receiver during deployment.



Multiple receivers can be associated with the same station name and a receiver can be associated with multiple stations.

Selecting a station name in the Detections tab (see section 4.1.2) will show all detections in the open database from that station, including those from multiple receivers.

It is possible to associate detections with a station after the data has been offloaded from a receiver, but the original VRL file will not contain any location information. This post-deployment editing is explained in section 2.4.2.



WARNING: DO NOT give multiple locations the same Station Name. This causes ambiguity in the database.

2.4.1 Add a New Station

To associate a receiver with a particular station before it is deployed, you must first use VUE to add a station name to your database. Follow the steps listed below to add a station.

STEP 1		
Open the <i>Add Station</i> window by either clicking on the "Add a new station record" icon in the Detections ribbon (shown below) or by right-clicking on the station list and selecting "Add Station" (shown at far right).	* All Detections Receivers (0) Transmitters (0) Station Impor Detect Delete All Station	Date Time tations

Continued...

STEP 2

Enter the location's unique name in the box labelled "Name". The station name will be used to identify the station in the selection tree and in the Station field of the database. This name should be unique from all other station names.

Optionally, enter the location's latitude and longitude in the boxes marked "latitude" and "longitude". There are various formats that will be recognized, and you may use any of the formats listed in the table below when inputting your data. Hemisphere must be indicated by either using +/- or by N/S/E/W.

Regardless of which format of "latitude" and "longitude" you choose to input, the default display setting is in +/- Degrees and will automatically convert to that format. This default display can be changed in the *Options* window (see section 7.4.1).

Format	Latitude	Longitude
	+44.64085	-063.67083
Degrees	44.64085° N	063.67083° W
	N44.64085	W063.67083
	44°38.451′ N	063°40.250' W
Degrees/Mins	N4438.451	W06340.250
	+4438.451	-06340.250
D	44°38′27.07″ N	063°40′14.98″ W
Degrees/Mins/Secs	N443827.07	W0634014.98
	+443827.07	-0634014.98



2.4.2 Edit a Station

Ũ

TIP: Hold the "Alt" key and type "248" on the number pad to enter the degree symbol (°).

It is possible to change the station association of a detection or group of detections. To do so, select the desired detections and "drag and drop" them onto the correct station name within the station selection tree. Remember that you can use Shift key to select a group of detections together. If you want to change the station for an entire VRL file, see section 5.2.1.

You can also edit a station name, the location, and/or the description associated with a station. When station information is changed, it will change for all data associated with the original station name. It's better to edit a station name rather than delete a station and then add a new station at the same location because the data association is lost when a station is deleted.

Edit a station rather than delete it and add a replacement.

Home Detections	Events	7 4 75	v 🏹 ∔	• éx	i/ 🔌 🜖	•)		
Detections	Files	Filte	rs	Stations	Transm	itters		
* All Detections					6664 detections			
Receivers (4)	Date	Time	Code Space	ID	Transmitter	Receiver	Station	
Transmitters (23) Stations (2)	2014-01-14	14:37:29	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
Stations (5) A North shore - beach	2014-01-14	14:37:56	A69-1303	48540	A69-1303-48540	VR4-UWM-250014	North shore - beach	
North shore - east	2014-01-14	14:38:01	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
🌡 North shore - west	2014-01-14	14:38:17	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
Imported Log Files (5)	2014-01-14	14:38:32	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
T Detection Filters (0)	2014-01-14	14:38:36	A69-1303	48540	A69-1303-48540	VR2W-119500		
	2014-01-14	14:38:47	A69-9002	7613	A69-9002-7613	VR2W-119500		
	2014-01-14	14:38:48	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:39:04	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:39:19	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:39:35	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:39:50	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:40:06	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:40:22	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	
	2014-01-14	14:40:32	A69-9002	7612	A69-9002-7612	VR2W-105684	North shore - east	
	2014-01-14	14:40:38	A69-9004	16384	A69-9004-16384	VR2W-105684	North shore - east	

Follow the steps below to edit an existing station's information.

Innovasea recommends making a copy of your VUE database prior to making significant edits, since database changes are irreversible. See section 2.2, Saving a database copy. STEP 1 STEP 2 Open the *Edit Station* window by either clicking on the "Add a new Highlight the station in the Stations station record" icon in the Detections ribbon (on left, below) or by section of the Selection Tree. right-clicking on the station in the list and selecting "Edit Station" * All Detections (on right, below). Receivers (4) 2010-12-21 14:49:43 Transmitters (23) A69-130 Þ Stations (3) 4 2010-12-21 14:49:49 A69-1303 Stations (3) North Shore 2010-12-21 14-49-56 🖕 Nor<u>t Ct</u> A60-130 🧯 North shore - beach Edit Station Alt+Enter North shore - east 🔓 Nor 303 🛓 North shore - west Importe Add Station 303 Stations Imported Log Files (5) 7 Detectio 30 **Delete Station** Del Detection Filters (1) 30 Delete All Stations 30 2010-12-21 14:00:20

STEP 3

Enter the location's unique name in the box labelled "Name". The station name will be used to identify the station in the selection tree and in the Station field of the database. This name should be unique from all other station names.

Optionally, enter the location's latitude and longitude in the boxes marked "latitude" and "longitude". There are various formats that will be recognized and you may use any of the formats listed in the table below when inputting your data. Hemisphere must be indicated by either using +/- or by N/S/E/W.

Regardless of which format of "latitude" and "longitude" you choose to input, the default display setting is in +/- Degrees and will automatically convert to that format. This default display can be changed in the *Options* window (see section 7.4.1).

Format	Latitude	Longitude
	+44.64085	-063.67083
Degrees	44.64085° N	063.67083° W
	N44.64085	W063.67083
	44°38.451′ N	063°40.250' W
Degrees/Mins	N4438.451	W06340.250
	+4438.451	-06340.250
	44°38′27.07″	063°40′14.98″
	N	W
Degrees/MINS/Secs	N443827.07	W0634014.98
	+443827.07	-0634014.98

😧 Edit Station		
Name :		
North Shore - e	ast	
Latitude:	Longitude:	
45°50'58.69"N	62°30'55.82W	
Description:		
		~
		-
		OK Cancel
	Alwavs use uniqu	ue station
	names (no two s	tations with
	the same name)	to avoid
	ambiguity in the	database.

Enter a description of the location in the "Description" box for your reference (optional).

Click "OK" when finished.

TIP: The degree symbol (°) can be entered on the keyboard by holding the "Alt" key while typing "248" on the number pad.

3. DATA ANALYSIS - THE BASICS

Data processing with VUE begins by collecting the data from the various receivers being used in a study. The data collection, or offloading, process is explained in detail in the various receiver user manuals. Some additional steps are: time correcting data files (section 3.1), removing false detections (section 3.2), and performing residency searches if desired (section 3.3).

3.1 Time Correction

What is time correction and why should you do it? Innovasea submerged receivers rely on crystal oscillators to keep track of time. Due to manufacturing variations, the frequency of the crystal oscillators varies slightly

between receivers. Each receiver's internal clock will drift – losing or gaining up to 4 seconds per day. This time drift can be caused by changes in temperature and/or variations in the oscillator. In most cases clock drift is highly linear and can be easily corrected.

We strongly recommend timecorrecting data <u>before</u> beginning any data analysis.

Time correcting receiver data aids in lining up detections by ensuring that detections from all receivers in a study are timestamped according to a common reference. For example, if five receivers receive the same transmission from Fish A, the time stamp of the detections can vary across receivers by several minutes or more. This variance may lead one to believe that multiple tag transmissions from this animal were detected on the receivers in the area; in reality, however a single transmission was detected at different times due to clock drift. Correcting this clock-drift would reveal detection of a single transmission at the same time on multiple receivers. Note this is an especially important consideration when attempting to assess directionality of animal movement, as in a survival study in a river system.

STEP 1

Establish that computer clock settings were correct (displayed time of day consistent with selected time zone) when the receiver was initialized and when the data were offloaded. Time correction is performed based on the assumption that the initialization and offload PC clock settings are correct. If either is incorrect then the time correction changes will be incorrect. Times can be edited if necessary (see Step 4).



STEP 2

Open the File menu and select Tools \rightarrow VRL File Editor.



Continued...

STEP 3

😢 Edit Files × VUE + Jan2014Data ← ← Search Jan2014Data P Select the VRL files you wish to time correct. Organize • New folder II • 🔲 📀 * * Name Date modified Туре You may edit multiple files at one time by holding the CTRL key while VR2W_100047_20101221_1.wl VR2W_105684_20140114_1.wl VR2W_119500_20140114_1.wl VR2W-RLD_105684_20140114_1.wl VR2W-RLD_119500_20140114_1.wl
 1/13/2014 1246 PM
 VRL File

 1/14/2014 1:54 PM
 VRL File
 selecting multiple files with the mouse. Click "Open". **C** a, 14 a- + -----Your default location will open automatically. File name: "VR2W_119500_20140114_1.vrl" "VR2W_1 See section 7.4.2 to change your default location. Open 🔽 Cancel STEP 4

ect the Auto Correct button in the VRL ing window.	VRL File Editing Options							
ing window.								
ing window.	File Name	Time Zone	Initializat	ion Time	Offloa	nd Time	Station	Import
	VR2W_100047_20101221_1.vrl	UTC	2013-12-21	14:47:51	2013-12-21	18:37:33	Isle Haute N	
0	VR2W_100047_20101221_1_edited.vrl	UTC	2013-12-21	14:47:51	2013-12-21	18:37:33	Isle Haute N	
	VR2W_105684_20140114_1.vrl	UTC	2013-11-18	14:57:02	2014-01-14	17:53:59	Isle Haute NW	
Auto Correct	VR2W_105684_20140114_1_edited.vrl	UTC	2013-11-18	14:57:02	2014-01-14	17:53:59	Isle Haute NW	V
	VR2W_119500_20140114_1.vr	UTC	2012-12-19	13:11:08	2014-01-14	17:48:58	Isle Haute E	
	Detais Auto Correct Res	et	2012-12-19	13:11:08	2014-01-14	17:48:58	Isle Haute E	Help
	When importing a VRL data set that irre	ady exists in t	he database:					
lotice that beneath each chosen file	Keep the existing data set (do not	mport VRI)						
a novu filo with " oditod" opponder i								
a new life with "_edited appended	f noodod rooot onv	ev	ver					
the filename Your existing VRI	i needed, reset any				i i		1	
	hanges made manual	lv 🛛	T	C	2010-	12-21	14:4	7:51
les will not be altered but a new				-				
dited file will be created and stored	or with the Auto Correc	τ -		С	2013-	11-18	14:5	7:02
ulteu lile will be createu allu storeu	utton to the dates tim	<u></u>	-	~	2012 1	1 10 -	44.5	7.02
/ith your existing VRL files.		03,		L .	2012-1	1-10 +	14:5	7:02
, ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	or time zones.	_	Т	C	2012-	12-10	12.1	1.08
				C	2012	12-15	15.1	1.00
			T.	<u> </u>	2012	12_10	12.1	1.00
a date, time, or time zone is incorrect, click nd make the necessary change(s). Make s een made.	< on the incorrect inform sure the Auto Correct b	mation outton	in the is seled	white cted <u>a</u>	line (eo <u>fter</u> the	dited f chan	îile) ge has	
When selecting Auto Correct if you	u notice a time drift of	morei	than 3() min	utes o	r if vo	u think	that
the original Time Zone and/or Office	and Time is incorrect,	please	contac	t Innc	vasea	Supp	ort.	unat

Continued...

STEP 5

Watch for very large differences in times between the offloaded time and the corrected offloaded time as this may indicate there's a problem.

If you're uncertain if the times in a VRL file are correct, then uncheck the "Import" box on the right to not import that VRL file into the database, look into the issue, and import the file later.

Click "Import" to save the edited files and import the corrected data into the open database.

Select what you would like done with any duplicate data that may be imported and click "Import".

RL He Ediung Opuons							
File Name	Time Zone	Initializat	ion Time	Offloa	d Time	Station	Import
/R2W_100047_20101221_1.vrl	UTC	2013-12-21	14:47:51	2013-12-21	18:37:33	Isle Haute N	11123
/R2W_100047_20101221_1_edited.vrl	UTC	2013-12-21	14:47:51	2013-12-21	18:37:34	Isle Haute N	
/R2W_105684_20140114_1.vrl	UTC	2013-11-18	14:57:02	2014-01-14	17:53:59	Isle Haute NVV	
/R2VV_105684_20140114_1_edited.vri	UTC	2013-11-18	14:57:02	2014-01-14	17:34:17	Isle Haute NVV	
/R2W_119500_20140114_1.vn		2012-12-19	13:11:08	2014-01-14	17:48:58	Isle Haute E	
	0.0		10111100			and reduce a	
Details Auto Correct Rese	et				7		Help
/hen importing a VRL data set that alre	ady exists in th	ne database:				_	
 Replace the existing data set Replace the existing data set only if 	the VRL is ne	5	shown	in red			
Replace the existing data set Replace the existing data set only if	the VRL is ne	5	shown	in red		Import	Canc
Replace the existing data set Replace the existing data set only if cont Receiver Log Files	the VRL is ne	5	shown	in red	×	Import	Canc
O Replace the existing data set Replace the existing data set only if port Receiver Log Files When importing a VRL data	the VRL is ne	s already ex	shown	in red	× :	Import	Canc
 Weplace the existing data set Replace the existing data set only if port Receiver Log Files When importing a VRL dat Weep the existing dat Replace the existing co Replace the existing co 	the VRL is ne ta set that a set (do 1 lata set lata set or	s already ex not import nly if the VI	shown kists in the VRL) RL is new	e database	:	Import	Cano

The three options are:

Keep the existing data set (do not import VRL)

This option will not import a VRL if that same VRL has already been imported into VUE and therefore does nothing with the selected file.

Replace the existing data set

This option will remove the previously imported VRL file before importing the selected file. Use this option if you want to replace a VRL file in the database with an edited version of the same file.

Replace the existing data set only if the VRL is newer

This option will remove the previously imported VRL file before importing the selected file but only if the selected file is newer than the previously imported one. Use this option if you have several edited versions of a file and you want to have the newest one in the database.

Duplicate detections are defined as multiple records of a detection made for the same transmitter, by the same receiver at the exact same time. If you import a VRL file multiple times, or import two or more VRL files which have overlapping data, for example, by offloading data from a receiver and redeploying the receiver without erasing the previous data, VUE will identify duplicate detections and remove them from the database.

None of these options will detect overlapping data sets. If a detection has already been imported from a different data set (either a subset or a superset), it will not be overwritten.

3.2 False Detection Analysis Tool

All communication systems experience transmission errors which can result in false detections. The Vemco system is quite conservative in its approach and thus the number of false detects is normally low, however they can still occur.

The **False Detection Analysis (FDA) tool** is designed to allow you to quickly identify the detections that may be questionable and therefore require further scrutiny. This tool does not state that the identified detections are false, but rather that the detections occurred in such a way that further analysis is needed before determining if the detection was false or not.

The FDA tool is based on an algorithm that uses the premise that false detections are generally separated by long intervals with only occasional short intervals. Details on the algorithm, *False Detections: What They Are and How to Remove Them from Detection Data*, can be found in the "Downloads" section of support.fishtracking.innovasea.com.

The FDA tool analyzes your detection data and determines, on a receiver- or station-to-tag basis, the number of short and long intervals between detections. Those tag-receiver or tag-station records for which there are more long intervals than short intervals are flagged as questionable data and require additional scrutiny before being accepted by the researcher as valid data. Single detections of a tag on a given receiver or at a given station are also flagged as questionable.

3.2.1 Using the FDA tool





STEP 5

Enter the desired start and end times within which to search. If desired, select a specific location (station or receiver) to limit the search area.

View available stations by selecting View by Station in the box to the right of the Location drop-down list.

View the list by receiver by selecting **View by Receiver**.

Choose a particular transmitter to search by selecting from the **Transmitter** drop-down list.

File	Home	Detections	Events	FDA Analyzer		
×	U					
	Search					
	Start Date	2000-01-01	-	00:00	•	(UTC)
	End Date	2015-10-16	-	00:00	•	(UTC)
	Location	All Stations			•	 View by Station View by Receiver
	Transmitter	All Transmi	tters		•	 View By Name View By Serial Number View By ID

You may perform the search on a transmitter's Name, Serial Number, or ID.

NOTE: Only transmitters that have been named will appear in the View by Name drop-down list.

Similarly, unless you specify a serial number for a transmitter, it will not appear in the **View by Serial Number** drop-down list.

For dual-ID sensor tags, it is best to perform the search either by *Name* or *Serial Number*; otherwise, the FDA tool assumes that each ID belongs to a different animal.

By default, the false detection analysis is performed on all locations and all transmitters.

STEP 6

Click the blue "Execute" button and wait for the results. Any tags with single detections - or that have more long intervals than short intervals between detections - on a given receiver or station are shown in red, and the acceptance is listed as "Questionable".

By default, VUE sorts FDA results by *transmitter*. To sort by another field, simply click on the column header. Click on the same column header a second time to reverse the sort order.

Hite Home	Detections Events	FDA Analyzer							•
Start Date	2000-01-01 -	00:00	• (UTC)						
End Date	2015-08-20 -	00:00	-		_				
Location	Any Receiver		Click any colur	nn heade	r to sort				
Toomittoo	Any Terrenitter		using that colu	mn. Click	again to				
Transmitter	Any Transmitter		get the reve	rse sort o	order.				
			41 tags spanning 6664 det	ections. 10 tags	s identified.				
ransmitter/	Receiver	Detections	Minimum Interval	Short Interv	als Long Intervals	First Detected	Last Detected	Acceptance	٦
	incourter a	Deceeding		Shore sheer					
469-1303-48535	VR2W-105684	45	312	39	0	2014-01-13 18	2014-01-14 13	Passed	1
A69-1303-48535 A69-1303-48540	VR2W-105684 VR2W-105684	45	312	39 161	0	2014-01-13 18 2014-01-13 18	2014-01-14 13 2014-01-14 16	Passed Passed	
A69-1303-48535 A69-1303-48540 A69-1303-48540	VR2W-105684 VR2W-105684 VR2W-119500	45 165 41	312 6 6	39 161 38	0 0 1	2014-01-13 18 2014-01-13 18 2014-01-13 18	2014-01-14 13 2014-01-14 16 2014-01-14 17	Passed Passed Passed	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540	VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014	45 165 41 39	312 6 6 7	39 161 38 30	0 0 1 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18	2014-01-14 13 2014-01-14 16 2014-01-14 17 2014-01-14 14	Passed Passed Passed Passed	
A69-1303-48535 A69-1303-48540 A69-1303-48540 469-1303-48540 469-1303-62364	VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014 VR2W-100047	45 165 41 39 25	312 6 6 7 34	39 161 38 30 24	0 0 1 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15	2014-01-14 13 2014-01-14 16 2014-01-14 17 2014-01-14 14 2010-12-21 15	Passed Passed Passed Passed Passed	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-62364 A69-1303-62366	VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014 VR2W-100047 VR2W-100047	45 165 41 39 25 55	312 6 6 7 34 6	39 161 38 30 24 54	0 0 1 0 0 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15 2010-12-21 14	2014-01-14 13 2014-01-14 16 2014-01-14 16 2014-01-14 17 2014-01-14 14 2010-12-21 15 2010-12-21 15	Passed Passed Passed Passed Passed Passed	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-62364 A69-1303-62366 A69-1303-65536	VR2W-105684 VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014 VR2W-100047 VR2W-100047 VR2W-100047	45 165 41 39 25 55 134	312 6 6 7 34 6 8	39 161 38 30 24 54 133	0 0 1 0 0 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15 2010-12-21 15 2010-12-21 14 2013-10-18 14	2014-01-14 13 2014-01-14 16 2014-01-14 16 2014-01-14 17 2014-01-14 14 2010-12-21 15 2010-12-21 15 2013-10-18 15	Passed Passed Passed Passed Passed Passed Passed	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-62364 A69-1303-62366 A69-1303-65536 A69-1601-12923	VR2W-105684 VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014 VR2W-100047 VR2W-100047 VR2W-100047 VR4-UWM-250014	45 165 41 39 25 55 134 1	312 6 6 7 34 6 8 0	39 161 38 30 24 54 133 0	0 0 1 0 0 0 0 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15 2010-12-21 14 2013-10-18 14 2013-12-09 12	2014-01-14 13 2014-01-14 16 2014-01-14 16 2014-01-14 17 2014-01-14 14 2010-12-21 15 2010-12-21 15 2013-10-18 15 2013-12-09 12	Passed Passed Passed Passed Passed Passed Passed Questionable	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-62364 A69-1303-62366 A69-1303-65536 A69-1601-12923 A69-1601-45220	VR2W-105684 VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014 VR2W-100047 VR4-UWM-250014 VR4-UWM-250014 VR4-UWM-250014	45 165 41 39 25 55 134 1	312 6 6 7 34 6 8 0 0	39 161 38 30 24 54 133 0 0 0	0 0 1 0 0 0 0 0 0 0 0 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15 2010-12-21 14 2013-10-18 14 2013-12-09 12 2013-12-10 18	2014-01-14 13 2014-01-14 16 2014-01-14 17 2014-01-14 14 2010-12-21 15 2010-12-21 15 2013-10-18 15 2013-12-09 12 2013-12-10 18	Passed Passed Passed Passed Passed Passed Passed Questionable Questionable	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-62364 A69-1303-62364 A69-1303-6536 A69-1601-12923 A69-1601-45220 A69-1602-8280	VR2W-105684 VR2W-105684 VR2W-105684 VR2W-105001 VR4-UWM-250014 VR2W-100047 VR4-UWM-250014 VR4-UWM-250014 VR4-UWM-250014 VR4-UWM-250014	45 165 41 39 25 55 134 1 1	312 6 6 7 34 6 8 0 0 0	39 161 38 30 24 54 133 0 0 0 0	0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15 2010-12-21 14 2013-12-09 12 2013-12-09 12 2013-12-10 18 2013-12-16 00	2014-01-14 13 2014-01-14 16 2014-01-14 16 2014-01-14 17 2010-12-21 15 2010-12-21 15 2013-12-10 18.15 2013-12-09 12 2013-12-10 18 2013-12-16 00	Passed Passed Passed Passed Passed Passed Passed Questionable Questionable	
A69-1303-48535 A69-1303-48540 A69-1303-48540 A69-1303-48540 A69-1303-62364 A69-1303-62366 A69-1303-62366 A69-1601-12923 A69-1602-8280 A69-1602-8280 A69-1602-22583	VR2W-105684 VR2W-105684 VR2W-105684 VR2W-119500 VR4-UWM-250014 VR2W-100047 VR4-UWM-250014 VR4-UWM-250014 VR4-UWM-250014 VR4-UWM-250014	45 165 41 39 25 55 134 1 1 1 1 1	312 6 6 7 7 34 6 8 0 0 0 0 0 0 0 0	39 161 38 30 24 54 133 0 0 0 0 0	0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2014-01-13 18 2014-01-13 18 2014-01-13 18 2014-01-13 18 2010-12-21 15 2010-12-21 14 2013-12-20 12 2013-12-20 12 2013-12-10 18 2013-12-16 20 2013-12-16 20	2014-01-14 13 2014-01-14 16 2014-01-14 16 2014-01-14 17 2014-01-14 14 2010-12-21 15 2013-12-21 15 2013-12-20 12 2013-12-10 18 2013-12-16 20 2013-12-16 20	Passed Passed Passed Passed Passed Passed Questionable Questionable Questionable	

3.2.2 Reviewing "Questionable" Data

It's important to note that FDA analysis does not conclusively determine that the "questionable detections" are false, only that the data require further scrutiny. Some things to consider when looking at questionable data are:

- 1. Detection activity. If there are many tags around a receiver, tag collisions can happen and can cause false detections. Single detections, or multiple detections spaced far apart recorded in the presence of other tags should be viewed very cautiously as they are most likely to be false.
- 2. Detections of a certain ID number on other receivers may provide you with enough confidence to classify a questionable detection as real, assuming the timing makes sense.
- 3. Fixed tags or range test tags if you have deployed tags for a range test and they are on the edge of the detection range, it will be normal for them to fade in and out during the range testing. Depending on the range performance these could look like detections that are separated by long intervals and thus flagged as questionable even though they are legitimate detections.
- 4. The short and long intervals are set up by default as 30 minutes and 12 hours respectively. This works well with normal tag delays (e.g. 30 sec to 5 minute delays). If your tags have very long random delays tens of minutes and longer you may not detect any short intervals and thus they are flagged as questionable even though they are valid detections. The default interval times can be changed as required (see below).
- 5. High Residency situation if you have a large number of tags resident around a receiver, you may expect to see longer intervals between tag detections due to tag collisions. It may be necessary to stretch out the minimal interval for the FDA analysis to allow for these longer intervals.
- 6. Finally, when in doubt call us. We are happy to have our team of experts review your data and provide guidance on whether to accept detections as real.

3.2.3 Changing FDA Interval times

There are situations when the FDA tool's default interval times may need to be adjusted according to the needs of your data. If you decide to do so, access to the FDA Parameters by opening the *Options* window (see section 7.4.6) and selecting the FDA Tool tab. The default times are shown in the window below.

Options								×
Display	Offload	Exp	ort	Devices	Releases	FDA Tool	Other	
FDA	Paramete	rs						
	Short inte	rval	30		minutes			
	Long inte	rval	12		hours			
						ОК		Cancel

3.3 Residency Search

The residency search option provides a method of reducing the volume of your data by summarizing detection counts by transmitter and receiver or station during a fixed period of time. You can specify the start and stop times and the number of detections required for a residency event (i.e. how many times must a tag be detected before it's considered in residence at a particular location). The "Absence Threshold" allows you to specify the maximum time between detections to be considered part of a single residency period at a location. Specific receivers or transmitters can be selected as parameters of the residency search.

The Residency Search is particularly helpful if you are working in a fixed area, such as a reef, and you wish to know when the tagged animals arrived in the area and when they left, but you are not interested in how many times they were detected while they were in the area.

Follow the steps below to use Residency Search:



Continued...

STEP 5

Enter the desired start and end times within which to search. If desired, select a specific location (station or receiver) to limit the search area. View available stations by selecting **View by Station** in the box to the right of the **Location** drop-down list. View the list by receiver by selecting **View by Receiver**.

Start Date	2000-01-01 🔹	00:00	UTC)
End Date	2015-08-19 💌	00:00	▼ (UTC)
Location	All Stations		 ♥ View by Station ♥ View by Receiver
Transmitter	All Transmitters		 View by Name View by Serial Number View by ID
Sort By	Arrival Time		•

Choose a particular transmitter to search by

selecting from the **Transmitter** drop-down list. You may perform the residency search on a transmitter's *Name*, *Serial Number*, or *ID*. Only transmitters that have been named will appear in the **View by Name** drop-down list.

Similarly, unless you specify a serial number for a transmitter, it will not appear in the **View by Serial Number** drop-down list.

For dual-ID sensor tags, it is best to perform the residency search either by *Name* or *Serial Number*, otherwise, the residency tool assumes that each ID belongs to a different animal.

By default, the Residency Search is performed on all locations and all transmitters.

The **Sort By** field lets you select if the results will be listed based on when a tagged animal *arrived* or when it *departed*.

STEP 6

Select the **Absence Threshold** (the maximum length of time permitted between detections within a single residency period), and the **Detection Threshold** (the minimum number of detections required for a residency event to be reported.)

The **Allow residencies to span time limits** option, when enabled, allows residencies to span the start and end time limits. When disabled, residencies are cut off before the limit.

Absence Threshold 1:00:00 🚔 (hh:mm:ss)						
Detection Threshold 4						
Allow residencies to span time limits						

Continued...

STEP 7	▼ Ross_Lagon-2012.vdb	STEP 8
Click the "Execute" button to begin the search. The results are listed in the lower section of the window.	File Home Detections Events Residency Search Search Search Image: Search Image: Search Start Date 2000-01-01 00:00 U(UTC) Absence Threshold 1:00:00 (hhrmmss) End Date 2015-08-20 00:00 U(UTC) Detection Threshold 2 Image: Search Location All Stations Image: View by Station Image: Allow residencies to span time limits Transmitter All Transmitters View by Station View by Juin	Export the search results as a CSV file by clicking the "Export search results as a CSV file" icon and entering a directory and file name.
	69 residencies spanning 6641 detections. 23 detections rejected.	
Search	Transmitter Receiver Arrival Departure Detections Absence (days) Ad9-1303-6286 VR2W-100047 2001-12:11 14:49:3 2010-12:11 14:49:7 55 Ad9-1303-6286 VR2W-100047 2001-12:11 14:49:17 55 5 Ad9-1303-62864 VR2W-100047 2001-12:11 14:49:17 2010-12:11 14:49:17 55 Ad9-1303-62864 VR2W-10047 2010-12:11 14:49:17 2010-12:11 14:49:17 55 Ad9-1303-62864 VR2W-100584 2013-10:18 14:49:27 2013-10:18 15:07:19 134 Ad9-3002-16284 VR2W-10584 2014-10:13 2016-03:11 11 2014-01:13 2016-03:11 2014-01:13 2016-03:11 Image: Separate date & time columns Image: Separate date & time columns Image: Transmitter Image: Transmitter Image: Transmitter	Search

STEP 9

Close the Residency tab by clicking the red "X" icon in the ribbon. More than one Residency tab can be open at the same time.



4. VIEWING DATA

Data are retrieved from receivers using the Offload feature. Specifics for offloading data are described in your receiver's user manual. If a database is open when the data are offloaded from a receiver, then a prompt will ask if the data should be imported into the open database. Data can be added to a database at any time by importing the .VRL file(s) into VUE.

4.1 Viewing Detections

The Detections tab contains all detections in the open database and is only visible if a database is open (see section 2.1 for opening a database). While in the Detections tab, it's possible to graph detections, search data for tag residency, import and export files, filter data, add/edit station or transmitter information, and view data by receiver, transmitter, or station.

_Lagon-2012.vdb						
Home	Detections	Events				
,		D . B				

		Nam	e of o	pen data	abas	e Seleo	ct "Detections	s" tab to			
The ribbon	Ross_Lagon-2012.vdb					see (letection info			x	
soction contains	Ele Home Detection							·		0	
	Home Detection						• • •				
functions that can		-	< T	. 7. 7	/ 🏜	L 🛶 🌡 /	••••••••••				
be performed with		- ·					······································				
the data such as	Detections	Files	Files Filters Stations Transmitters								
	* All Detections		6664 detections								
graphing or	 Transmitters (23) 	Date	Time	Code Space	ID	Transmitter	Receiver	Station	Data		
filtering (see	Stations (3)	2014-01-13	19:05:08	A69-9002	7614	A69-9002-7614	VR2W-105684	North shore - east	3 ADC		
soction $(1, 1, 1)$	Imported Log Files (5)	2014-01-13	19:05:10	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
Section 4.1.1)	P Tetection Filters (1)	2014-01-13	19:05:26	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
		2014-01-13	19:05:31	A69-9002	16264	A69-9002-16264	VR2W-119500		20.00 °C		
		2014-01-13	19:05:42	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
		2014-01-13	19:05:58	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
		2014-01-13	19:06:14	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
		2014-01-13	19:06:24	A69-9002	7612	A69-9002-7612	VR2W-119500		4 ADC		
Selection Tree:		2014-01-13	19:06:30	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
Select "All Detections", a receiver, a		2014-01-13	19:06:46	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
		2014-01-13	19:07:02	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
transmitter, a station, a log file, or a		2014-01-13	19:07:05	A69-9002	7613	A69-9002-7613	VR2W-105684	North shore - east	155 ADC		
filter to display the associated data in		2014-01-13	19:07:18	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
		2014-01-13	19:07:35	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
the right side of the VUE window.		2014-01-13	19:07:51	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
-			19:08:06	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC		
Each item in the list can be expanded		2014-01-13	19:08:22	A69-9004	16384	A69-9004-16384	VR4-UWM-250014	North shore - beach	3 ADC	-	
so an individual receiver, transmitter								3 1	8:39:08 UT	Ci	
eta con ha viewad (see section 4.4.0)		1								1	
etc. can be viewed (see section 4.1.2).											
						γ					

The right side of the VUE window changes according to what's selected in the tree on the left side of the window. If a receiver is selected, then this section of the window reports which tags were detected by that receiver. If a tag is selected, <u>all</u> the detections in the database for that tag are shown. If a station is selected, all detections in the database recorded at that station are shown.

4.1.1 Detections Ribbon

Selecting the Detections tab will bring the Detections Ribbon to the fore, allowing any of the available features to be selected. Each feature is explained elsewhere in the manual – see the references below.



4.1.2 Detections in the Selection Tree

The selection tree on the left side of the window allows access to the detections of individual receivers, transmitter, stations, imported log files, and detection filters. Clicking on the \triangleright symbol next to a category will open the expanded view, as shown below.



4.1.3 Viewing All Detections

To view all detections in a database, first select the Detections tab at the top of the window. This tab is only visible when a database is open. Select "All Detections" at the top of the selection tree on the left hand side of the VUE window. All detections, sorted by date, will be listed on the right-hand side of the screen.



Time

Date

4.1.4 Viewing Detections by Receiver

Receivers are uniquely named according to their model and serial number, for example, VR2W-100282. To view detections collected with a specific receiver, select the receiver in the selection tree on the Detections tab. All detections from that receiver, sorted by date, will be listed on the right-hand side of the screen.

4.1.5 Viewing Detections by Transmitter

To view detections of a specific transmitter, select the transmitter in the selection tree on the Detections tab. All detections from that transmitter, sorted by date, will be listed on the right-hand side of the screen.

4.1.6 Viewing Transmitter Detections

Code Space

Each transmitter is provided an ID number at the factory. The transmitter data sheet which was shipped with the transmitter will provide the transmitter serial number, the transmitter code space, and the ID number. Within VUE, the default transmitter name will be displayed as a combination of the code space and the ID number. A Code Space is a term used to describe the type of message transmitted by a tag. It is important to note that tags with the same ID number but different code spaces (e.g., A69-1601-2056 and A69-9001-2056) do NOT represent the same tagged animal. It's important to include both Code Space and ID (ex: A69-1601-2056, referred to as the "full ID") when analyzing detection data to avoid any ambiguity in identifying your tagged animals.

By selecting the transmitter name in the list on the left side of the VUE detections window, VUE will show all detections found in the current database for that transmitter.

example: A69-1601-

Receiver

Default Transmitter Name: code space – ID#

Transmitter

2020-11-13 16:29:17 A69-1601 62011 A69-1601-62011 VR2Tx-483190 2020-11-13 16:29:26 A69-9007 11236 A69-9007-11236 VR2Tx-483190 2020-11-13 16:30:01 A69-9007 11236 A69-9007-11236 VR2Tx-483190 11236 2020-11-13 16:30:31 A69-9007 A69-9007-11236 VR2Tx-483190

ID

For more information on receiver Code Maps and transmitter Code Spaces, please visit the **Code Spaces and Code Maps** topic at <u>support.fishtracking.innovasea.com</u>.



🌾 Receivers (10)



36
4.1.6.1 Transmitter name

A transmitter can also be assigned a user-defined name. To do this, highlight the tag in the selection tree and click the *Edit selected transmitter* in the Transmitter section of the Detections ribbon (shown at right), or right-click on the tag in the selection tree and select *Edit transmitter*. Either method will open the *Edit Transmitter* window shown below. The serial number and a user-defined name can be entered to be assigned to that tag.

The user-defined transmitter name will appear in the "Transmitter" column on the right hand side of the VUE window when "All Detections", "All Events" or a receiver, station, log file, or filter is selected on the left hand side. The database columns "Code Space" and "ID" will still contain the code space and transmitter ID.

4.1.7 Viewing by Station

As receivers are mobile units easily moved from one location to another, VUE has provided the option to associate a receiver with a station (location) for a particular deployment. The station specifies a location, including latitude and longitude (if desired), which is assigned and recorded in the receiver's memory during initialization. More information on stations is found in section 2.4.

A station is used to identify the location of the receiver during deployment.

nitter" l
Tag S/N:
Tag S/N:
Tag Name:
OK
Cancel





×

4.2 Plotting Data

Any data displayed in the Detection tab can be plotted by time. The plot can be viewed by selecting the "Chart Detections" icon in the Detections ribbon (shown at right). The detections plotted are selected using the selection tree on the left. Selecting a transmitter will plot all the detections from that transmitter, grouped by receiver. Selecting a receiver will plot all the detections from that receiver, grouped by transmitter. Selecting "All Detections" will plot all database detections, grouped by transmitter.

Adding or removing data from the database automatically updates the plot.

There are two plot display modes in VUE: a detail mode and an overview mode. When you first generate a plot, the data are shown in Detail mode. In this mode, the plot window is capable of simultaneously displaying just under 8, 200 detections. The number of detections currently displayed, as well as the total number of detections in the selected data are shown on the top left of the plot window. The black section of the grey bar at the top of the plot indicates which portion of your selected data you are currently viewing. In the example below, the plot shows the first 199 of all 1,224 detections for the receiver VR2W-119500.



Detections can be grouped by transmitters, receivers, or stations. The legend on the right hand side of the window provides a list of the transmitters, receivers, or stations for which detections are currently visible on the plot.



To return to detail mode, click on

the "Detail" button at the bottom of the plot window.

For large datasets, overview mode may take considerable time to display. For this reason, VUE provides the option to abort loading the overview plot. To do so, click on the "Abort" button at the bottom left of the plot window.

To view all detections in your database simultaneously, click on the "Overview" button in the navigation tools (below the chart). Overview mode will display all detections but will not display the details of each detection.



To more easily distinguish the time of day each transmission was detected, you can opt to select the Day/Night Shading box above the chart. When selected, a dialog box initially appears where you can set the sunrise and sunset times or use latitude and longitude values. The plot will then show any night detections with a shaded area, as shown in the sample below.

The Chart tab contains several navigation tools to help browse your data. These are listed and described in the table below.

Button	Description
<	Go to the beginning of the data selection
>	Go to the end of the data selection
(or keyboard left arrow)	Skip back one display page (keyboard left arrow)
(or keyboard right arrow)	Skip forward one display page (keyboard right arrow)
Zoom In (or keyboard down arrow)	Display a shorter time range. You can also zoom in on a particular area by clicking anywhere on the graph and dragging to define a time range.
(or keyboard up arrow)	Display a longer time range (keyboard up arrow)
Overview	Display entire dataset at once

In both detail and overview modes, moving the mouse pointer over any point on the plot will display the transmitter ID number, the receiver it was stored on, and the date and time details of that detection (see example at right).



4.2.1 Visualizing predation events

For tags in the VUE database that have been identified as predation sensors (refer to Section 2.3.3, *Predation Sensor Tags*), the Detections chart allows visualization of the state of the sensor tag at the time the tag was detected. A predation sensor tag is represented in the chart by a square as long as it is in either an "untriggered" or "triggering" state. Once the tag has triggered and begins its count of time since triggering, tag detections are represented by a hexagon.



4.3 Filtering Data

As well as allowing you to view detection data by receiver, transmitter, station, or log file, VUE provides the capability to view your data by detection filter. You can create detection filters to look at subsets of your data that meet particular criteria such as date, time, transmitter ID, transmitter frequency, receiver, station, number of detections, and transmitter/receiver combination. For example, if you are interested in looking at all of the detections in your database for transmitter Tag ID #3 that occurred *after* March 5, 2006 at 2:00 pm and excluding detections at "Big Red Rock" station, then VUE makes it easy to create a filter with those definitions. Detection filters provide an easy way to view, plot, and export subsets of your database.

4.3.1 Creating Detection Filters

To create a detection filter, follow the steps listed below.

- 1. Select the Detections tab.
- 2. Select the *Add Detection Filter* icon (shown at right) in the Detections ribbon or right-click on "Detection Filters" in the selection tree and choose "Add Filter". The "Add Detection Filter" window will appear (shown below).
 - Add Detection Filter
 Image: Second state of the following conditions:

 New Filter
 will pass all detections that
 Imatch all Image: of the following conditions:

 Detection Date/Time
 is after
 YYYY-MM-DD, hh:mm:ss

 OK
 Cancel
- 3. Enter the filter name in the box labelled "New Filter".
- 4. Choose either "Match all" or "Match any" from the drop-down list to the right of the New Filter box. If you filter your data using more than one criterion you can control whether the data must match all of your chosen criteria or at least one of the chosen criteria.
- Using the drop-down menu in the box labelled "Detection Date/Time", choose your first criterion. VUE provides thirteen different options for filter criteria. These criteria are listed and described in section 4.3.3.
- 6. In the next drop-down menu box, choose your desired "qualifier" for the filter criterion that you have selected. The qualifiers available will depend on the filter criterion that you have selected. For example, if you choose to filter by Detection Date/Time, the available qualifiers are "is after" or "is before". Refer to section 4.3.3 for a complete list of qualifiers for each filter option.
- 7. In the box on the bottom right of the Add Detection Filter window, type the desired data value to use in the filter.



- 8. To add another filter criterion to the same filter, click on the + sign on the right-hand side of the window. A new row will appear for you to enter the next filter criterion.
- 9. When you are finished adding filter criteria, click OK.

Filters that have been created are shown in the selection tree under "Detection Filters". Your filtered data can be viewed by expanding the detection filters branch and clicking on the filter of interest in the selection tree. The data are displayed in the detection window on the right hand side of the screen. When your data are displayed, they can be plotted by selecting the *Chart data* icon (shown at right) in the Detections tab.

To export your filtered data, select the *Export File* icon under the Detections tab (shown at right). Metadata associated with your filtered detections can also be exported, by selecting *Export Metadata*. Details on exporting and importing metadata are found in section 5.8.2.

4.3.2 An Example of a Common Filter

How do I create a filter that queries all detections of a transmitter with two IDs (such as a TP tag) between two dates?

Step 1: Create a filter that displays all detections for two IDs.

Edit Detection Filter		
Tag ID	will pass all detections that of the following conditions:	
Transmitter	 ▲ A69-9002-7611 ▲ A69-9002-7612 ▲	'n
	OK Cancel	

Step 2: Create a filter that displays all detections for 2 IDs between 2 dates using the filter created in Step 1 as a prefilter.

ate & Tag ID Code		will pass al	l detections that	match all	of th	e following	conditions:
Prefilter Detection Date/Time	passesis after	*	Tag ID	▼ 0:00:00	- + - +	A	II condition must be met!
Detection Date/Time	▼ is before	• •	2013-09-01, 0	0:00:00	- +		
						ОК	Cancel





4.3.3 Filter Options

VUE provides thirteen different options for filtration criteria that you can use to build your detection filters. The table below lists and describes the options.

Filter Criteria Options				
Filter Criterion	Qualifiers	Description		
Detection Date/Time	 is after is before	Displays all detections in the database occurring after (or before) a given date and time. If no time is specified, the default is 00:00:00 on the date given.		
Transmitter	• is • is not	Displays all detections from (or not from) a given transmitter selected from a list of all transmitters in the current database. For example, "A69-1303-9" (transmitter with a default label) or "Salmon 58" (a user-named transmitter).		
Transmitter Name	 is is not contains does not contain begins with does not begin with ends with does not end with 	Displays all detections of transmitters matching or not matching the specified Transmitter Name, or portions of the Transmitter Name. This criterion only searches <i>named</i> transmitters, i.e. transmitters that have been manually named in VUE. Transmitters that have not been assigned a name other than the default will be ignored by this filter. To filter by the default transmitter label assigned by VUE (e.g. "A69-1303-9"), use the "Transmitter" filtration criterion above.		
Transmitter Code Space	• is • is not	Displays all detections from (or not from) the specified Coding ID. Transmitter Code Space refers to the last 4 digits found in the "Code Space" column of the VUE window for each transmitter. For example, to find all detections for transmitters in the 1601 code space, set the Transmitter Code Space filter to "is" and its value to "1601".		
Transmitter Tag ID	 is is not is greater than is less than 	Displays all detections for transmitters matching, not matching, greater than, or less than the specified Transmitter Tag ID. The transmitter tag ID is shown in the "ID" column of the VUE window.		
Transmitter Frequency	 is is not is greater than is less than 	Displays all detections for transmitters of the specified frequency. The transmitter frequency is contained in the first 3 characters if the "Code Space" column in the VUE window. For example, if the Code Space is A69-1601, the transmitter frequency is 69 kHz.		
Transmitter Detection Count	 is is not is greater than is less than 	Displays all detections from any transmitter for which the database contains (or does not contain) a given number of detections. You can also search for transmitters with detection counts above or below a given value.		

Filter Criteria Options				
Filter Criterion	Qualifiers	Description		
Receiver	• is • is not	Displays all detections recorded by (or not recorded by) a given Receiver selected from a list of all Receivers in the current database.		
Station	• is • is not	Displays all detections that occurred at (or not at) a given Station, selected from a list of all Stations in the current database.		
Station Name	 is is not contains does not contain begins with does not begin with ends with does not end with 	Displays all detections at Stations matching or not matching the specified Station Name, or portions of the Station Name		
Rxr/Txr Detection Count	 is greater than is less than is is is not 	Displays all detections from any receiver/transmitter pair for which the database contains (or does not contain) a given number of detections. You can also search for receiver/transmitter pairs with detection counts above or below a given value.		
Rxr/Txr Detection Separation	is less thanis greater than	Displays all detections that are within less than or greater than a given number of minutes of another detection from the same receiver/transmitter pair.		
		Displays all detections that pass or do not pass an existing filter in the database. The Prefilter provides a way to string together a series of filters that you have already created for your database. It is the only way to combine filters that use the "Matches all of" and "Matches any of" options.		
Prefilter	• passes • does not pass	For example, you have defined a filter called "Recent Salmon Detections" that returns all detections from Transmitter Names containing "salmon" AND occurring after March 6, 2007 (using the "Matches all of" option), and another filter called "Shallow Stations" that returns all detections that occurred either at Station 1 OR at Station 2 (using the "Matches any of" option). Now you can use the Prefilter option to create a third filter called "Recent Salmon Detections, Shallow" that returns all the detections that pass both your "Recent Salmon Detections" and "Shallow Stations" filters.		

4.3.4 Editing and Deleting Detection Filters

is not

is

Ŧ

•

Add Detection Filter

Station Name

Transmitter

Filter 6

To edit an existing filter, select the *Edit Filter* icon in the Detections ribbon (select Detections tab t "Edit to Fi er options to be

will pass all detections that

Ŧ

Ŧ

match all

edited.	
lter". The Edit Detection Window, shown below, will appear to allow any	of the filte
see Detections ribbon), or right-click on the filter name in the selection tree	e and selec
bedit an existing filter, select the <i>East Filter</i> icon in the Detections ribbon (s	select Dete



of the following conditions:

To delete a filter, select the Delete Filter icon in the Detections ribbon, or right-click on the filter name in the selection tree and select "Delete Filter." You may also delete all filters in your database by selecting "Delete All Filters" after right-clicking on a filter name in the selection tree.

4.3.5 Exporting Detection Filters to Another Database

Filter definitions can be exchanged between databases using the Metadata Import and Export options. To transfer existing filters from one database to another, follow the steps listed in Section 5.8.2, "Transferring Metadata between Databases."



×

5. MANAGING DATA

5.1 Importing Data

A VRL file (see section 1.1.2) is created every time data are offloaded from an Innovasea receiver. Choosing not to automatically import the data into a database at the time of offloading can save you time in the field. If this is the case, or if you need to import a VRL for another reason, then follow the instructions below. VRL files can be imported into any number of databases depending upon your needs.

Reduce retrieval time in the field by offloading data files first and importing into the database later.

Note that importing a VRL file from anywhere other than the default directory will cause that file to be copied into the default



Innovasea **recommends** making a copy of your VUE database prior to making significant edits, since database **changes are irreversible**. See section 2.2, **Saving a database copy**.

directory. This ensures that you always have a local copy of every imported file. This feature of the VUE software is enabled by default, but can be disabled by un-checking "Copy imported VRL files into this folder" on the Offload tab of the *Options* window (see section 7.4.2 to open *Options*).

STEP 1

Select the Import file icon on the Detections tab. If the Detections tab is not visible, open or create a database (see section 2.1).



STEP 2

Select one or more VRL files to import and click "Open". Multiple files can be selected by pressing and holding the CTRL key while left clicking on each desired file.

The VRL files selected in the example shown here have all been time corrected before being imported into the database. This procedure, found in section 3.1, is strongly recommended by Innovasea.

Hold the CTRL key to select multiple files.



VRL files with the **RLD** identifier in the filename (RLD data file, as in "VR2AR-RLD_545565_20141216_1.VRL") cannot be imported into VUE.

The files	are ir	nmediately	imported	into th	e database	e (single	file impo	rt shown).

ĺm	porting VRL Files
	Successfully imported VR2W_100047_20101221_1.vrl
	ОК

If you attempt to import one or more VRL files that have already been imported into VUE, VUE will ask whether you want to replace the existing file(s) with the one(s) that you are importing. This may happen if, for example, you are replacing a VRL file in the database with an edited version of the same file. Click **Yes** to replace the file(s), and **No** to keep the file(s) that are already in the database.

Warning if importing a single file that already exists

STEP 3

Import Receiver Log Files	
VR2W_100047_20101221_	1.vrl already exists in the database. Do you want to replace it with VR2W_100047_20101221_1.vrl?
	Yes No

Warning if importing multiple files that already exist

Import Receiver Log Files	J
One or more of these files already exist in the database. Do you want to replace them?	
Yes No	

Note: If you import a VRL file multiple times, or import two or more VRL files which have overlapping data (for example, by offloading data from a receiver and redeploying the receiver without erasing the previous data), VUE will identify duplicate detections and remove them from the database.

Continued...

STEP 4

Check the results to verify the files imported as expected. The results will be different depending on what you selected in Step 3. For example, selecting "NO (do not import VRL)" option will give us the results on the left while using the "YES (you want to replace them)" option gives us the results on the right. The same VRL files were imported into the same database but the example of the left reported that two "error" messages alerting us to the fact that duplicate VRL files were not imported.

File Processing Results		×	File Processing	g Results		×
<pre>** ERRORS OCCURRED ** Successfully Imported 2 of 4 files VRL Import Results</pre>	Data set was previously ERROR - Duplicate Data Data set was previously ERROR - Duplicate Data SUCCESS SUCCESS	imported imported	VRL Impor VR2W_1000 VR2W_1056 VR2W_1195 VR4UWM_21 Print	ct Results 	.vrl : SU .vrl : SU .vrl : SU ed.vrl : SU	CCESS CCESS CCESS CCESS
Print Save As		ОК				

5.2 Editing a VRL file

VRL files are a record of the information that was offloaded from a receiver. It's important that these original records be kept unaltered so that the validity of the original data remains intact. For this reason, when editing VRL files a new VRL file is created and the original file is left unaltered. Edited VRL files are labelled with "_edited" appended to the file name.

Original VRL files are never altered – edits are saved in a new VRL file with "_edited" added to the name.

Why would we need to edit a VRL file? The most common reason is to correct detection data for time drift (Time Correction); this is explained in detail in section 3.1. Correcting for an incorrect PC time during initialization, and/or to add or correct a station name are also reasons to edit a VRL file.

STEP 2

Select the VRL files you wish to edit.

You may edit multiple files at one time by holding the CTRL key while selecting multiple files with the mouse.

Click "Open".

Your default location will open automatically. See section 7.4.2 to change your default location.



🔵 🖡 « Ve	mco 🕨 VUE 🕨 Jan2014Data	✓ 4 Search Jan20.	14Data 🕽
ganize • Ne	v folder		. 0
A Name	*	Date modified	Туре
VR2	W_100047_20101221_1.vrl	1/13/2014 12:46 PM	VRL File
VR2	W_105684_20140114_1.vrl	1/14/2014 1:54 PM	VRL File
VR2	W_119500_20140114_1.vrl	1/14/2014 1:48 PM	VRL File
VR2	W-RLD_105684_20140114_1.vrl	1/14/2014 1:54 PM	VRL File
VR2	W-RLD_119500_20140114_1.vrl	1/14/2014 1:48 PM	VRL File
L.			
2			
2			
_ +		111	
File name:	VR2W_119500_20140114_1.vrl= "VR	2W_1 - Receiver Log Files	(*.vrl) •

STEP 3

Click on the incorrect information in the white line, corresponding with the edited file, and make the necessary change(s). This example shows an incorrect offload time being adjusted.

The existing VRL file is not altered but a new edited file will be created and stored with your existing VRL files.

View the details of all the selected files.



Continued...

STEP 4

Select the edited files that will be imported into the open data base. If a file is unselected (no checkmark), the edited file will be saved in the selected directory (section 7.4.2) but the data will not be imported into the database. It can be imported at a later time (section 5.1).

File Name	Time Zone	Initializat	ion Time	Offloa	d Time	Station	Import
/R2W_100047_20101221_1.vrl	UTC	2010-12-21	14:47:51	2010-12-21	18:37:33	North shore - beach	
VR2W_100047_20101221_1_edited_1.vrl	UTC	2010-12-21	15:47:51	2010-12-21	19:37:33	North shore - beach	1
VR2W_105684_20140114_1.vrl	UTC	2013-11-18	14:57:02	2014-01-14	17:53:59	North shore - west	
VR2W_105684_20140114_1_edited_1.vrl	UTC	2013-11-18	14:57:02	2014-01-14	17:45:59	North shore - west	
VR2W_119500_20140114_1.vrl	UTC	2012-12-19	13:11:08	2014-01-14	17:48:58		
VR2W_119500_20140114_1_edited_1.vrl	UTC	2012-12-19	13:11:08	2014-01-14	17:48:58	North shore - east	1
Edi	ted dat own in	a are red.					
When importing a VRL data Sh	•••••						

STEP 5

Select how you want VUE to handle any data conflicts upon import. These options are explained in section 3.1, Step 5.

Click the "Import" button to make the changes to the VRL files and import the selected files into the open database.



5.2.1 Correcting or Adding a Station Name to a VRL file

To change or add a station name to a VRL file, follow the instructions in section 5.2 and select the correct station name from the pull down menu under Station in the row(s) for the edited VRL file(s). You will only be able to select from a list of existing station names within your VUE database. If you need to add additional station names, you will need to exit this window and add a new station.

North shore - beach North shore - beach North shore - west North shore - west	V
North shore - beach North shore - west North shore - west	
North shore - west	
North shore - west	
North Shore West	V
-	
North shore - west	
North shore - beach North shore - east	Help
	Vorth shore - west North shore - beach North shore - east

5.3 Managing Duplicate Tags in VUE

Note: The following section provides guidance on managing a database with duplicate tag ID numbers. This can occur if you combine data from multiple years or multiple users. This is typically only an issue for older transmitter code spaces sold prior to 2014 (ex: A69-1105) due to the limited numbers of unique IDs that these code spaces provided. Modern transmitter code spaces provide many more unique IDs and do not require ID duplication. If you don't expect your database to have duplicate tag ID numbers, you can skip this section of the manual.

5.3.1 What is a duplicate tag?

If you are comparing older detection data from multiple years, multiple studies, or from multiple users, then it's possible that your data sets will contain duplicate ID numbers for different tags.

Duplicate ID numbers will often be separated in the database by date or location and therefore it is usually straightforward to distinguish one transmitter from another. For sensor tags, it is important to ensure that tag detections are associated with the correct sensor tag record within the VUE database. The correct sensor calibration (e.g., slope & intercept) information must be associated with each transmitter.

When a VRL data file is imported into the VUE database, VUE inspects each detection for a previous record of the ID number in the database. If a matching tag record exists in the database, the new detection is associated with that existing tag. If more than one tag exists in the database matching the detection's tag ID, the detection is associated with the *first* matching record in the database. If the new detection is the first entry of a given tag in the VUE database, then a new tag record is created using the tag type and ID number contained in the detection.

If you edited a sensor transmitter record to include a serial number and sensor calibration information, the raw sensor data (ADC) will be converted into meaningful units (e.g. degrees Celsius). If a transmitter database record has a serial number, then any new detections imported will be highlighted and marked as an *Unverified Transmitter*. This is a warning that the detection has been associated with an existing transmitter serial number and the association may be wrong if there are duplicate tag ID numbers within the dataset.

Two examples for managing duplicate tag ID numbers follow. The first is an example where duplicate tags are not present in the data, and the second is an example with duplicate tag ID numbers in the same dataset.

listed below at left.

Example – No Duplicates

	Date	Time	Code Space	ID	Transmitter	Receiver	Station	Data
A69-1105-1	2008-02-12	13:32:51	A69-1105	1	A69-1105-1	VR2W-100037	Ben's Brook	199 ADC
A69-1105-2	2008-02-12	13:33:05	A69-1105	2	A69-1105-2	VR2W-100037	Ben's Brook	0 ADC
A69-1303-3	2008-02-12	13:34:26	A69-1105	1	A69-1105-1	VR2W-100037	Ben's Brook	198 ADC
A69-1303-1	2008-02-12	13:34:40	A69-1105	2	A69-1105-2	VR2W-100037	Ben's Brook	0 ADC
A69-1303-3	2008-02-12	13:36:54	A69-1105	1	A69-1105-1	VR2W-100037	Ben's Brook	194 ADC
A69-1303-9	2008-02-12	13:37:08	A69-1105	2	A69-1105-2	VR2W-100037	Ben's Brook	0 ADC
	2008-02-12	13:38:49	A69-1105	1	A69-1105-1	VR2W-100037	Ben's Brook	192 ADC

In the following case, we've created a new database and offloaded data from two receivers: a VR2 with serial

number 7731 and a VR2W with serial number 100037. The detections from the receivers belong to the transmitters

The detections for transmitters A69-1105-1 and A69-1105-2 are for sensor tags, so we add sensor tag calibration information. This converts the values in the data column from "ADC" (Analog-to-Digital Converter) units to measurement units (e.g. meters or degrees Celsius). The calibration information (slope and intercept) is found on the tag datasheets sent by Innovasea with the sensor tags.

To calibrate, follow the instructions in Section 2.3.2 (*Edit Information for an Existing Sensor Tag*). Right-click on the selected transmitter in the selection tree list and select "Edit Transmitter" from the list that appears.

Transmitters (6)	4)		Date	Time
s	Edit Transn	nitter		Alt+Enter
s) A69-1. s) A69-1.	Add Transr Add Duplic	nitter ate Tr	ansmitter	
s•) A69-1	Delete Tran	smitt	er(s)	Del
A69-1200-1	100		2006-10-19	02:27:17

In the *Edit Transmitter* window, fill in the serial number, slope, and intercept from the datasheet provided by Innovasea and select the sensor type.

The data are converted to °C and all the detections will be highlighted and flagged as "Unverified Transmitter" as shown below. The highlighting is a warning that the detection(s) may or may not be associated with the correct transmitter record.

Date	Time	Code Space	ID	Receiver	Station	Data		
2008-02-12	13:32:51	A69-1105	1	VR2W-100037	Ben's Brook	14.726 °C		
2008-0	Ununviliad	Tunn conittou				.632 ℃		
2008-0 Thic	D8-Q							
2008-0 The a	association m	hay need to be cor	rrected i	f duplicate tag ID n	umbers are present	067 °C		
2008-02-1	13:40:33	A69-1105	1	VR2W-100037	Ben's Brook	13.973 °C		
2008-02-12	13:43:33	A69-1105	1	VR2W-100037	Ben's Brook	13.879 °C		
2008-02-12	13:46:15	A69-1105	1	VR2W-100037	Ben's Brook	14.161 ℃		



and the warning will disa

3. Drag and drop the detections onto the

correct transmitter name under the

transmitter selection tree. The detections	2008-02-12	13:34:26	A69-1105	1	VR2W-100037	Ben's Brook
	2008-02-12	13:36:54	A69-1105	1	VR2W-100037	Ben's Brook
will now be intentionally associated with	2008-02-12	13:38:49	A69-1105	1	VR2W-100037	Ben's Brook
this transmitter record and the highlighting	2008-02-12	13:40:33	A69-1105	1	VR2W-100037	Ben's Brook
and the warning will disannear	2008-02-12	13:43:33	A69-1105	1	VR2W-100037	Ben's Brook
and the warming will disappear.	2008-02-12	13:46:15	A69-1105	1	VR2W-100037	Ben's Brook

Time

2008-02-12 13:32:51 A69-1105

Code Space ID

1

Receiver

VR2W-100037

Station

Ben's Brook

Data

14.726 °C

14.632 °C

14.255 °C

14.067 °C

13.973 °C

13.879 °C

14.161 °C

NOTE: If these detections are deleted from the database and re-imported from the VRL, the detections will need to be verified again.

In this case, we know that all of these detections are from one animal. Therefore, we can verify these detections are associated with the correct (and only) transmitter data record with this ID number. To verify an ID number

2. Select all the detections for that transmitter in the right hand VUE window by pressing and holding the

Example – Duplicate IDs

and its data:

How to separate duplicate ID numbers (different fish)

5.3.2 Verifying a transmitter and its data

SHIFT key and clicking on the last detection in the list.

1. Select the desired transmitter in the selection tree.

In this example, we have duplicate detections for sensor tag A69-1105-2. Most of these detections were offloaded from the receiver when it was at station "Ben's Brook" and the rest were offloaded from the same receiver when it was at "Big Red Rock". The ID 2 at "Ben's Brook" was determined to be a depth transmitter and the ID 2 detected at "Big Red Rock" was determined to be a temperature sensor. If we apply the temperature sensor

calibration information to the depth sensor, all the detections will be converted to degrees Celsius. This would be an error, as some of these detections belong to another fish with a different sensor.

In order to apply calibration

information separately, create a second

Date Time Code Space ID Receiver Station Data VR2W-100037 2008-02-12 2 Ben's Brook 13:33:05 A69-1105 0 ADC 2008-02-12 13:34:40 A69-1105 2 VR2W-100037 Ben's Brook 0 ADC 2 2008-02-12 13:37:08 A69-1105 VR2W-100037 Ben's Brook 0 ADC 2008-02-12 13:39:03 2 VR2W-100037 A69-1105 Ben's Brook 0 ADC 2008-02-12 13:40:45 A69-1105 2 VR2W-100037 Ben's Brook 0 ADC 2008-02-12 2 Big Red Rock 13:46:08 A69-1105 VR2W-100037 0 ADC 2 2008-02-12 13:46:29 A69-1105 VR2W-100037 Big Red Rock 0 ADC

transmitter record and associate the "Big Red Rock" detections with the new record.

To do this, add a duplicate transmitter record. Transmitter records are considered duplicates if their Code Space and ID are identical, in this case, A69-1105-2. The serial number and calibration information will be different.

Select the transmitter from the selection tree and right click the mouse and select "Add Duplicate Transmitter".

📢 Receivers (11)	^				
🌒 Transmitters (6	4)	Date		Time	Co
s A69-1105-1	Edit Transmiti	:er	A	t+Enter	A6
s) A69-110 s) A69-110	Add Transmitter				
s) A69-110		e Transm	litter -		A6
 A69-110 A69-1206-T 	Delete Transr 100	nitter(s)	De - 10- 19	el 02:27:17	A6
A69-1206-1	102	2006	10 10	02.27.40	

The newly created transmitter has no detections associated with it. All of the tag ID A69-1105-2 detections within the database are currently associated with the first A69-1105-2 record.

Next, select the "Big Red Rock" temperature sensor detections and drag and drop them onto the new transmitter record in the selection tree. This will remove them from the first transmitter record and associate them with the second transmitter record.

We can now edit each of the two transmitter records to add serial number, calibration information, sensor type etc. We suggest that you change the display name of the transmitters to help differentiate them.

5.4 Removing files from VUE

Data that were imported from a VRL file but are no longer wanted in a database can be removed. Removing a VRL file from a VUE database does not remove it from its storage location on your computer's hard drive. The VRL file is unchanged and remains available for import at a later time.

To remove a file, it must be selected in the selection tree on the left side of the window when the Detections tab is

selected (below). This enables the *Remove file* icon in the Detections ribbon (right). A file can also be removed by right-clicking on the desired file name in the selection tree and selecting *Remove file* in the list.

NOTE: All detections associated with the VRL file will be removed from the VUE database.

File Users Detertions Surets							6
Per Home Detections Events	Eilterr	-	tation	•	Transmitters		
* All Detections			Stations	1309 det	ections		
Keceivers (4)	Date	Time	Code Space	ID	Transmitter	Receiver	
Transmitters (23)	2013-11-27	14:10:59	A69-1303	36829	A69-1303-36829	VR2W-105684	1
 Stations (3) Imported Log Files (5) 	2013-11-27	14:11:06	A69-1303	36829	A69-1303-36829	VR2W-105684	
Unknown Device 0 21490607 2.vrl	2013-11-27	14:11:12	A69-1303	36829	A69-1303-36829	VR2W-105684	
VR2W_100047_20101221_1.vrl	2013-11-27	14:11:19	A69-1303	36829	A69-1303-36829	VR2W-105684	
VR2W_105684_20140114_1.vrl	2013-11-27	14:12:46	A69-1303	36829	A69-1303-36829	VR2W-105684	
VR2W_119500_20140114_1.vrl	2013-11-27	14:12:53	A69-1303	36829	A69-1303-36829	VR2W-105684	
VR4UWM_214906_20140114_1_edited_2.vrl	2013-11-27	14:12:59	A69-1303	36829	A69-1303-36829	VR2W-105684	
p Detection ritters (1)	2013-11-27	14:13:06	A69-1303	36829	A69-1303-36829	VR2W-105684	
	2013-11-27	14:13:12	A69-1303	36829	A69-1303-36829	VR2W-105684	
	2013-11-27	14:13:19	A69-1303	36829	A69-1303-36829	VR2W-105684	
	2013-11-27	14:13:25	A69-1303	36829	A69-1303-36829	VR2W-105684	
	2013-11-27	14:13:38	A69-1303	36829	A69-1303-36829	VR2W-105684	
	2013-12-04	14:05:01	A69-9002	6819	A69-9002-6819	VR2W-105684	
	2014-01-13	18:12:17	A69-9002	16264	A69-9002-16264	VR2W-105684	
	2014-01-13	18:12:25	A69-9002	16264	A69-9002-16264	VR2W-105684	
	2014-01-13	18:12:33	A69-9002	16264	A69-9002-16264	VR2W-105684	•
						🚯 18:53:10 U	лс

Removing a VRL file from a VUE database does not remove it from its storage location (VRL file) on your computer's hard drive.

All detections associated with the removed VRL file will be removed from the VUE database.



Change the display name of the duplicate transmitters to help differentiate them in the selection tree.

5.5 Removing Transmitters from VUE

In some cases, a database may contain detections that have been analyzed and identified to be *false*. False IDs or other transmitters that are no longer desired in a database can be removed by deleting the transmitter(s).

Removing a transmitter from VUE will remove all detections associated with the transmitter. While deleted transmitters can be restored by re-importing the source VRL file or files into the database, it is recommended that you make a copy of your VUE database before performing any deletions (see section 2.2, Saving a database copy). This precaution can simplify restoring deleted data if it is determined at a later date that some of the removed detections were valid.

To delete one or more transmitters, select them in the selection tree on the left side of the Detections window and click on the "Delete the selected transmitter record" icon in the Detections ribbon (right). You may also right-click on any selected transmitter in the selection tree and then choose "Delete Transmitter(s)".



To select more than one transmitter for deletion, hold the CTRL key while selecting multiple transmitters with the mouse.

File Home Detection	ons E	Events						
🚮 · 🔎 · 🛃		-	7	% 7/	4	∲x ∳∕ ∮	🎍 🦦 🥠	
Detections	F	iles	1	Filters		Stations	Transmitters	
* All Detections	^					245 detections		
Receivers (4) Transmitters (23)		Date	Time	Code Space	ID	Receiver	Station	D.
 A69-1303-23541 		2014-01-13	18:19:37	A69-1303	48540	VR2W-105684	North shore - east	
A69-1303-36829		2014-01-13	18:19:56	A69-1303	48540	VR4-UWM-250014	North shore - beach	
A69-1303-48535		2014-01-13	18:20:36	A69-1303	48540	VR2W-119500		
A69-1303-48540	=	2014-01-13	18:25:52	A69-1303	48540	VR4-UWM-250014	North shore - beach	
A69-1303-62364		2014-01-13	18:26:32	A69-1303	48540	VR2W-119500		
 A69-1303-62366 A60 1202 65526 		2014-01-13	18:38:51	A69-1303	48540	VR2W-105684	North shore - east	
A69-1601-12923		2014-01-13	18:39:10	A69-1303	48540	VR4-UWM-250014	North shore - beach	
A69-1601-45220		2014-01-13	18:39:50	A69-1303	48540	VR2W-119500		
A69-1602-8280		2014-01-13	18:44:49	A69-1303	48540	VR2W-105684	North shore - east	
A69-1602-22583		2014-01-13	18:45:07	A69-1303	48540	VR4-UWM-250014	North shore - beach	
A69-1602-33977		2014-01-13	18:45:47	A69-1303	48540	VR2W-119500		
 A69-1602-46020 A60-1602-60044 		2014-01-13	18:51:58	A69-1303	48540	VR4-UWM-250014	North shore - beach	
A69-1602-50944 A69-1602-54019		2014-01-13	18:58:37	A69-1303	48540	VR4-UWM-250014	North shore - beach	
s) A69-9002-6819		2014-01-13	19:09:54	A69-1303	48540	VR2W-105684	North shore - east	
s) A69-9002-7611		2014-01-13	19:15:34	A69-1303	48540	VR2W-105684	North shore - east	
s) A69-9002-7612	-	2014-01-13	19:21:51	A69-1303	48540	VR2W-105684	North shore - east	
	F.	2014-01-13	19:27:19	A69-1303	48540	VR2W-105684	North shore - east	

A Warning box appears, reminding you of the delete action you are about to perform. Verify that the transmitter number displayed is the one you want to delete, then click **Yes**.



NOTE: When multiple transmitters have been selected, the dialog box will ask: "*Are you sure you want to delete the selected transmitters*?"

5.6 Viewing Millisecond Timestamps

Detection data timestamps with millisecond precision can be exported from VUE and viewed if the VRL files were created with a VR2W receiver using firmware 3.0 or greater. To export millisecond timestamp information, follow the steps below.

Deterr with m 1. 2. 3. 4. 5.	STEP 1 mine if the .VRL file(s) can be exported nillisecond timestamps: View the .VRL file in VUE Right-click on the file name Select "Inspect file" Check for the FW value in the file information listed – version 3.0 or greater Close the File Processing Results window	File Processing Results VREW_114311_20130415_1.vrl Original File Name : Source Flug-in : Receiver Model : Serial Number : Version : Receiver Time § Log Sart : Receiver Time § Log Sart : Receiver Time § Log Upled : PC Time § Ind; Log Upled : PC Time § Log Upled : Decoding Map : Blanking Interval :	VR2W_114311_201300415_1.vrl 2013-02-12 VR2C69_450000_20130027_2.vrl 2013-02-12 VR2C69_450008_20130927_2.vrl 2013-02-12 VR2W_114311_2 Inspect File VR2W_114311_2 Remove File VR2W_114311_2 Remove File VR2W_114311_2 2013-02-12 VR2W_114311_2 Remove File VR2W_114311_12 Remove File VR2W_114311_2 Remove File VR2W_11431_2 Remove File VR2W_11431_2 Remove File VR2W_11431_2 <
Repea	at for any other files involved	Print	
Enable 1. 2. 3. 4. 5.	STEP 2 e millisecond exporting (see also section 7.4 Open the <i>Options</i> window Select the Export tab Select "Custom" from the list on the left Click "Millisecond time precision" if a checkr already present Click "OK"	I.3) nark is not	Options Image: Control of the control of
	STEP 3		

Export a CSV file (see section 5.7 for details).



5.6.1 Viewing millisecond timestamps in MS Excel

The CSV file contains the millisecond timestamps, but if you choose to open the file in MS Excel then an adjustment must be made to the cell formatting in Excel for the milliseconds to become visible. To make this adjustment, select the "Date and Time (UTC)" column and open the Format Cells window (right-click, select "Format Cells..."). Setup a custom numbering format by selecting "Custom" and entering **m/d/yyyy h:mm:ss.000** in the Type line. Select "OK".

Viewing the CSV file in an ASCII reader, such as Notepad, will display the millisecond times without requiring a formatting adjustment.

Number	Alignment	Font	Border	Fill	Protection		
Category General Number Currency Accounti Date Time Percenta Fraction Scientific Text Special Custom	: /ng ige	Sampl Date Type: m/d/y Gener 0 0.00 #,## \$#,#: \$#,#:	e and Time (l /yy h:mm:s al #0_);(\$#,# #0_);[Red](JTC) s.000 #0) (\$#,##0)			H
Type the	number forma	r code, usi	ng one of ti	ne existing	codes as a start	ing point.	Delete

5.7 Exporting Data

Data can be exported from a VUE database for use in other applications. VUE provides two format options for exporting detection data: comma-separated value (.csv) format and legacy VR2 format.

CSV files can be read by most spreadsheet, database and statistics applications; therefore, this export format should be used if you intend to analyze the detection data in another application. Alternately, you may wish to export your data from VUE in Legacy VR2 format similar to what was provided by VR2PC software.

To export detection data, select the detections you wish to export using the selection tree on the left hand side of the window (Detections tab or Events tab). Choose the Export file icon in the ribbon (shown below) and select your desired export format. Selecting CSV or VR2 formats will open the Data Export window, shown at right (VR2 format is not available in the Events tab). Note that regardless of the time zone display option you have selected in VUE, all detection data will be exported in UTC time, unless your Export options are configured to also export in display time (see Custom format below).



Data Export
Output File C:\Users\ltaylor\Documents\Ve\VUE_Export.csv Change
File Format
Omma-Separated Value (CSV)
CSV export options are configured through the global options window (Main Menu -> Options -> Export).
Legacy VR2 Format
Data will be converted to conform with the selected map and any detections that do not fit the map will be discarded. Times are in UTC.
MAP-110 (69 kHz)
Sensor Format
Calibrated sensor values
Raw sensor values
OK Cancel

To set your CSV file format export options, open the *Options* window (see section 7.4.3) and select the Export tab. The following describes the various formats available when exporting CSV files:

- > Default Format: Output detection data are in the default CSV export format (recommended setting).
- Version 1.0 Format: Output detection data are in the original CSV format. Use this setting for compatibility with older versions of VUE (prior to Version 1.8).
- **Custom Format**: Output detection data uses custom settings:
 - Column names may be based on the default or Version 1.0 CSV format
 - A column can be added for the time zone currently displayed in VUE. This column is in addition to the UTC time column, which is always output.
 - Date and time may be split into separate columns.
 - Spaces may be eliminated from column names. If in doubt, disable this option.
 - Trailing commas may be eliminated for rows. If in doubt, enable this option.
 - The output file may use ANSI or UTF-8 Unicode character encoding. Changing this option may eliminate garbage characters when viewing the file in an external application. If in doubt, select UTF-8 Unicode.

Options		×
Display Offload Export Devices	Releases FDA Tool Other	
Comma Separated Value (CSV))	
Default format	Custom Configuration	
O Version 1.0 format	Version 1.0 column names	
Oustom	☑ Include VUE date & time (currently	UTC)
	Separate date & time columns	Encoding
	No spaces in column headers	O UTF-8 Unicode
	No trailing commas	ANSI
	Millisecond time precision	
		OK Cancel

If you are exporting your data in Legacy VR2 format, you will need to select a Code Map to complete the process. In the old VR2PC format, the code map setup information is defined in the header of each file. To be compatible with the VR2PC format, the legacy format can only be supported for code maps with 4 types of coded tags. If the set of data, you have chosen contains more than four types of tags then you will not be able to export all of them into a single VR2 text file. You will get a warning describing the data that were not exported. Choose the remaining detections and an alternate map that is suitable for these detections to export the remainder of your data.

5.8 Metadata

5.8.1 What is Metadata?

Metadata consist of any calibrations and configurations that you have manually entered into VUE, such as transmitter sensor calibrations, transmitter names, stations names and locations, and filter definitions. Transferring metadata between databases that contain many of the same receivers, transmitters, and stations can save you appreciable time since you do not have to manually re-enter your definitions into the new database. In addition, because the information is transferred by VUE, you eliminate the possibility of introducing new typographical errors into the database that is receiving the information.

5.8.2 Transferring Metadata Between Databases

To transfer metadata to a new database, you will first need to export it from your existing database. Follow the steps below to **export** metadata.

- 1. Select the Detections tab and then the Export file icon (shown at right).
- 2. Choose Export Metadata from the drop-down list.

Metadata Evnort

- 3. Enter the desired file name for your metadata in the "Export File Name" window and click Save.
- 4. Check the boxes next to the items that you wish to export when the Metadata Export window appears. If you select the "Export Filters" option, then all receiver, transmitter and station metadata are also exported by default.

X Metadata Export

Export Transmitters
Export Stations
Export Filters

5. Click OK. A message box will appear indicating that your metadata were successfully exported.

VUE exports metadata in a custom XML file format (*.vxm). The default folder for VUE metadata files is "My Documents\Vemco\VUE\" when VUE is first opened. Every subsequent time, it will be the folder you last exported to.

Next, you will need to **import** the metadata into your target database. Follow the steps below to import metadata.

- 1. Open the target database for import (see section 2.1 for how to open a database).
- 2. Choose the Import file icon from on the Detections tab (shown at right).
- 3. In the "Import Files" window, locate the desired metadata file. Select the file and click "Open".



×I



4. Select the item definitions that you wish to import when the Metadata Import window appears.

By selecting "Import Filters" you will automatically include all Metadata Import × receiver, transmitter and station information. Metadata Import Options 5. If you have altered the metadata file manually, VUE will report ✓ Import Receivers that the checksum is invalid and not import the file. You can ☑ Import Transmitters override this by selecting "Ignore Invalid Checksums" Import Stations Import Filters 6. Click OK. 🔲 Ignore invalid checksums Select "Import Filters" to

automatically include all receiver, transmitter and station information.



The VXM Import Results window will appear, showing the number of successfully imported station, receiver, transmitter, and filter records as well as the number of rejected duplicate and incomplete records. Duplicate records are records for things that already exist in the database. Incomplete records are for things that were not completely defined in the exporting database. For example, transmitters without defined serial numbers are incomplete.

6. VIEWING EVENTS

Innovasea receivers record when certain events occur to the receivers, such as when a new study was started (initialized), or when data were offloaded. These events are shown in the VUE Events Tab.

6.1 Events Tab

The VUE event log describes a receiver's operational history over the span of a study. Each event records receiver status information for a time period.

If a database is open, an Events tab will be visible at the top of the window. To see the events for all the receivers combined, select "All Events" at the top of the selection tree (above right).

If a particular receiver is selected in the selection tree (left side of window), then the events related to that receiver or VRL file are listed on the right side of the window (right).

6.1.1 Grouping Events by Type

To view events grouped by event type, right click anywhere on the right side of the event window and select "Group Events by Type" (below right).

To return to default sorting behaviour (by Date and Time), right click in the window again and deselect "Group Events by Type".

File Home Detections	Events				
Files			,		
* All Events				1797 events	
Keceivers (4)	Date	Time	Receiver	Description	Data
VR2W-100047	2010-12-21	14:47:51	VR2W-100047	Initialization	
VR2W-103084	2010-12-21	14:47:51	VR2W-100047	PC Time	2010-12-21 10:47:51 UTC-04:00
K VR4-UWM-250014	2010-12-21	14:47:51	VR2W-100047	Map	MAP-112 (A69-1105; A69-1206; A69
Imported Log Files (5)	2010-12-21	14:47:51	VR2W-100047	Blanking	260 ms
	2010-12-21	14:47:51	VR2W-100047	Station	Vemco
	2010-12-21	14:48:01	VR2W-100047	Daily Pings	0
	2010-12-21	14:48:01	VR2W-100047	Daily Syncs	0
	2010-12-21	14:48:01	VR2W-100047	Daily Rejects	0
	2010-12-21	18:37:28	VR2W-100047	Daily Pings	644
	2010-12-21	18:37:28	VR2W-100047	Daily Syncs	81
	2010-12-21	18:37:28	VR2W-100047	Daily Rejects	0
	2010-12-21	18:37:28	VR2W-100047	Daily Detections on A69	80
	2010-12-21	18:37:28	VR2W-100047	Last Detection on A69-1303	2010-12-21 15:43:07 UTC
	2010-12-21	18:37:33	VR2W-100047	Data Upload	VR2W 100047 20101221 1.vrl
	2010-12-21	18:37:33	VR2W-100047	PC Time	2010-12-21 14:37:34 UTC-04:00
	2012-12-19	13:11:08	VR2W-119500	Initialization	
	2012 12 10	12,11,02	VP2W/ 110500	BC Time	2012-12-10 12:11:08 LITC-04:00

Noss_Lago	1-2012.000				
E Home Det	ections	E	vents		
Files					
* All Events				15 events	
Keceivers (4)	Date	Time	Receiver	Description	Data
VR2W-100047	2010-12-21	14:47:51	VR2W-100047	Initialization	
VR2W-105084	2010-12-21	14:47:51	VR2W-100047	PC Time	2010-12-21 10:47:51 UTC-04:00
VR4-UWM-250014	2010-12-21	14:47:51	VR2W-100047	Map	MAP-112 (A69-1105; A69-1206; A69-1303; A
Imported Log Files (5)	2010-12-21	14:47:51	VR2W-100047	Blanking	260 ms
	2010-12-21	14:47:51	VR2W-100047	Station	Vemco
	2010-12-21	14:48:01	VR2W-100047	Daily Pings	0
	2010-12-21	14:48:01	VR2W-100047	Daily Syncs	0
	2010-12-21	14:48:01	VR2W-100047	Daily Rejects	0
	2010-12-21	18:37:28	VR2W-100047	Daily Pings	644
	2010-12-21	18:37:28	VR2W-100047	Daily Syncs	81
	2010-12-21	18:37:28	VR2W-100047	Daily Rejects	0
	2010-12-21	18:37:28	VR2W-100047	Daily Detections on A69	80
	2010-12-21	18:37:28	VR2W-100047	Last Detection on A69-1303	2010-12-21 15:43:07 UTC
	2010-12-21	18:37:33	VR2W-100047	Data Upload	VR2W 100047 20101221 1.vrl
	2010-12-21	18:37:33	VR2W-100047	PC Time	2010-12-21 14:37:34 UTC-04:00

Ross_Lagon-2012.vdb					
Home Detections	Events				
Files				1797 events	
Keceivers (4)	Date	Time	Receiver	Description	Data
Imported Log Files (5)	2010-12-21	14:47:51	VR2W-100047	Initialization	
	2010-12-21	14:47:51	VR2W-100047	PC Time	2010-12-21 10:47:51 UTC-04:00
	2010-12-21	14:47:51	Select All	Ctrl+A	MAP-112 (A69-1105; A69-1206; A.
	2010-12-21	14:47:51	Conv Selectio	n Ctrlv C	260 ms
	2010-12-21	14:47:51	Group Events	hy Type	Vemco
	2010-12-21	14:48:01	Gloup Events	by type	0
	2010-12-21	14:48:01	VR2W-100047	Daily Syncs	0
	2010-12-21	14:48:01	VR2W-100047	Daily Rejects	0
	2010-12-21	18:37:28	VR2W-100047	Daily Pings	644
	2010-12-21	18:37:28	VR2W-100047	Daily Syncs	81
	2010-12-21	18:37:28	VR2W-100047	Daily Rejects	0
	2010-12-21	18:37:28	VR2W-100047	Daily Detections on A69	80
	2010-12-21	18:37:28	VR2W-100047	Last Detection on A69-1303	2010-12-21 15:43:07 UTC
	2010-12-21	18:37:33	VR2W-100047	Data Upload	VR2W 100047 20101221 1.vrl
	2010-12-21	18:37:33	VR2W-100047	PC Time	2010-12-21 14:37:34 UTC-04:00
	2012-12-19	13:11:08	VR2W-119500	Initialization	
	2012-12-19	13-11-08	VR2W-119500	PC Time	2012-12-19 13:11:08 UTC-04:00

6.2 Events Ribbon

Clicking the Events tab will bring the events ribbon to the front, allowing you to select from the icons shown.



6.3 Events in the Selection Tree

The selection tree on the left side of the window allows access to the events of individual receivers and imported log files. Clicking on the \triangleright symbol next to a category will open the expanded view, as shown below.



6.4 Events Common to all Receiver Types

6.4.1 Reset Event

Reset events occur when the receiver internal software initializes and are described as "Normal Restart". This will occur if the battery is replaced or possibly if the battery connection is loose. Other Reset Events include "WATCHDOG RESET", "FLASH ACCESS VIOLATION" or "UNIDENTIFIED RESET". If you see any of these events, please contact Innovasea as they may indicate a receiver malfunction.

6.4.2 Initialization Event

Initialization events are created each time your receiver is initialized. When you import your VRL files into VUE, you will see the following information displayed in the Events tab for each receiver initialization.

Description	Data
PC Time Zone	Time zone of PC used to initialize the receiver, referenced to UTC time (e.g. UTC-04:00)
Мар	Displays Code Map and associated Code Spaces used for receiver initialization
Blanking	Receiver blanking time. This is the time for which the receiver stops listening after receiving a detection (to avoid detecting echoes)
Station	Station name entered during initialization
Study Description	Study Description text entered during initialization

6.4.3 Memory Capacity Event

The Memory Capacity displays amount of receiver memory, as a percentage, that has been used. For VR2W receivers, this is recorded each day along with all other receiver status information. For VR2Tx and VR2AR receivers, this is recorded when the data are offloaded from the receiver. For VR4-UWM receivers, memory capacity used is recorded with all other status information at the interval selected at the time the study was initialized.

6.4.4 Receiver-Specific Data Summaries

Innovasea receivers generate detection data summaries, which can be found in the Events log, at specific intervals depending on the receiver type. These events are stored along with the detection data in the receiver and are added to the VUE database when the VRL file is imported.

For VR2W, VR2Tx, and VR2AR receivers (with firmware version 5.0.1 or later), data summaries are recorded every hour. For firmware versions prior to 5.0.1, VR2W, VR2Tx, and VR2AR receivers record data summaries every 24 hours at midnight UTC. An entry containing a "partial hour" (or "partial day") of summary data is also recorded when data are offloaded. These "partial" data summary entries can be identified by noting the time the summary was created.

Data summaries are recorded every hour and when data are offloaded from a VR2W, VR2Tx, or VR2AR.

Receiver-Specific Data Summaries						
	VR	2W, VR2T	k, VR2AR*			
Data Summary Field		Data Summary Logging				
	Descriptor	Every hour	Every 24 hours (midnight UTC)	On data offload (partial interval)		
Battery level	Battery		~	v		
Number of pings received during time interval	Hourly Pings / Daily Pings / Pings / Study Pings	~	~	>		
Number of detections logged during time interval	Hourly Detections	~				
Number of syncs logged during time interval	Daily Syncs / Syncs		~	>		
Number of rejects (check sum errors) received during time interval	Daily Rejects / Rejects		~	✓		

*The above table refers to log files from receivers using firmware version 5.0.1 or later. Log files from receivers using earlier firmware versions may have different data summary events.

VR4-UWM receivers record data summaries at an interval selected at the time the study was initialized. VR4-UWM data summaries can be stored at an interval of between 1 to 14 days; the default is 1 day (24 hours).

6.5 Exporting Events

To export Event logs for use in other applications, first select the Events tab and then select a receiver or "All Events" in the selection tree on the left hand side of the VUE window. Next, select the "Export file" button (shown at right) in the Events ribbon near the top of the window and choose to export the information as either a CSV file or as a Metadata file. Event times are exported in UTC, regardless of the time zone display option selected in VUE.



VUE Events format got you down? Try converting your VRL files to CSV using our Fathom Connect software (visit the Downloads section of <u>support.fishtracking.innovasea.com</u> or our online Export tool at <u>fathomcentral.com</u>. The new Fathom CSV format contains detection and events data in a single file and is designed for easy parsing.

7. ADDITIONAL INFORMATION

7.1 Installing the VUE software

The VUE software requires either Windows VISTA, Windows 7, Windows 8, or Windows 10.

NOTE: *Bluetooth* is required for communication with the VR2AR, VR2Tx, VR2W, and VR4-UWM receivers. Use of your computer's built-in *Bluetooth* adapter is recommended, although a USB *Bluetooth* dongle may be used if required. See section 7.2 for details.



7.2 Connecting with Bluetooth

There are two ways to connect your PC or laptop through Bluetooth for communication with your receiver. Use the method that suits the equipment you have.

7.2.1 Using INTERNAL Bluetooth

There are a number of Windows Operating Systems (OS) PCs, laptops, and tablets equipped with **internal Bluetooth** capable of communicating with Innovasea receivers through the VUE software. Use of internal **Bluetooth** is strongly recommended where possible.

IMPORTANT: If you are using internal Bluetooth, **DO NOT INSERT** an external Bluetooth adapter into your device. Doing so may cause Bluetooth driver conflicts resulting in inability to communicate with your receiver.

- 1. Open VUE software (latest version available from support.fishtracking.innovasea.com).
- Observe the Bluetooth icon at the bottom of the window. In most cases, it will look like the Bluetooth icon shown at right, indicating that you're ready to communicate with Innovasea Bluetooth receivers. If VUE is not able to perform Bluetooth communication, you will see a red "X" through the icon.

NOTE: Communication difficulties may occasionally occur when the red "X" is not present. If this happens, please contact Innovasea Support at <u>support.fishtracking.innovasea.com</u>.

3. Connect to the receiver. Instructions are found in the receiver's user manual. If a Bluetooth connection cannot be established with the receiver, try using an external Bluetooth USB adapter as described in the next section.

7.2.2 Using the EXTERNAL Bluetooth Adapter

If your computer does NOT have internal Bluetooth capability, use an external USB Bluetooth adapter, as described here.

Attach the Bluetooth adapter (may not be as shown) to any USB port on your computer.

Follow Steps 1 – 3 above.

7.3 Auto Updates

If you see the VUE Software Update window (shown here), select "Yes" to install the new version immediately. If you select "No", the version of VUE that is currently installed on your PC or laptop will open and you will be prompted again the next time you open VUE.







7.4 Select VUE Options

There are options in VUE that can be changed according to your preference, accessed through the *Options* window.

Select the File tab and click the "*Options*" button at the bottom of the menu box to open the *Options* window.

File			
4	New Database		
2	Open Database		
	Save Database As		
×	Close Database		
3	Tools +		
1	About VUE		
		Options	📲 Exit VU

7.4.1 Display Settings

How the latitude/longitude, the date and time, and the sensor units are displayed in VUE is set in the *Display* tab of the *Options* window. The options are found in the corresponding drop-down lists. There is also an option to select which time will be displayed in VUE, the UTC (Coordinated Universal Time), the time with a fixed offset, or the local time. If you choose to view the local time, be sure to set the time on your PC/laptop to the correct local time, using a reliable time source such as a GPS device or a timekeeper website.

Options		
Display Offload Export Devices Releases F	DA Tool Other	
Latitude/Longitude	Date & Time Display	Date & Time Display
Degrees (ISO6709)	yyyy-mm-dd 🔻	yyyy-mm-dd 🔻
e.g: +44.64085-063.67083	UTC (Coordinated Universal Time)	UTC (Coordinated Universal Time)
Sensor Units	◎ Fixed UTC Offset: -04:00 ▼	● Fixed UTC Offset: -04:00 ▼
Depth: Metres	Clock Set Clock	Cocal time:
Pressure: Pascals		20 Holds, condar-connect and accounts from - Determit Diplome 20 Holds, condar-connect and accounts from - Determit Diplome Contract, Top 20 Holds, contract, Status, Sta
Acceleration: m/s ²		Local time in Halifax
		1 • 25 • 01 DN
	OK Can	

7.4.2 Offload Settings

Before offloading any data from your receivers, you may wish to change the directory location to which the data are offloaded. To change the storage folder, select the *Offload* tab in the *Options* window and click the "Change" button in the Receiver Offload Storage Folder section. The data files created when data are offloaded from a receiver will be stored in this new storage location. If this location is not changed, then the default directory is created at ... *Documents**Vemco**VUE**ReceiverLogs*\ and the data files will be stored there.

Offload file format can be selected from the choices listed in the Offload File Format section: VRL files only, VRL files and CSV files, or VRL files and legacy VR2 files. The option to additionally offload the raw data files from VR2W receivers is also found in this tab.

> Keep this checked as these files contain diagnostic information that may be useful in the future.

Options	
Display Offload Export Devices Releases FDA Tool Other	
Receiver Offload Storage Folder	
C:\Users\Itaylor\Documents\Vemco\VUE\ReceiverLogs\	Change
Copy imported VRL files into this folder	
Offload File Format	
VRL files only	
VRL files and CSV files	
VRL files and legacy VR2 files	
Additionally offload raw data file from VR2W receivers (RLD)	
L	
	OK Cancel

7.4.3 Export Settings

The format a CSV file takes when it is exported from VUE (see section 5.7) is determined in the Export tab of the *Options* window. The three options are the default format, the Version 1.0 format, and the custom format.

Detection data timestamps with millisecond precision can be exported from VUE and viewed if the VRL files were created with a VR2W receiver using firmware 3.0 or greater. The ability to export this millisecond precision is available using the Custom format. Instructions are located in section 5.6.

Options Display Offload Export Devic Comma Separated Value (CS)	Select if the date and time exported is local or UTC in the Display tab (section 7.4.1).	
Default format	Custom Configuration	
Version 1.0 format	Default 🔹 column name	s
 Custom 	Include VUE date & time (curren	tly UTC)
	Separate date & time columns	Encoding
	No spaces in column headers	UTF-8 Unicode
	No trailing commas	O ANSI
	Millisecond time precision	
		OK Cancel

7.4.4 Devices

The Devices tab in the *Options* window offers the option of automatically closing Bluetooth communications with a receiver if communication with that receiver has been inactive for 30 minutes. This is a power-saving feature

since more receiver battery power is needed to keep the communication open that is used for detecting and recording tags.

Options	x
Display Offload Export Devices Releases FDA Tool Other	
Receivers Automatically close inactive receiver connections after 30 minutes	

7.4.5 FDA Tool Settings

When using the FDA tool (see section 3.2.1), the interval parameters can be altered depending on the needs of your data. The default values are shown here.

Ор	tions							×	J
)isplay	Offload	Expo	ort	Devices	Releases	FDA Tool	Other	
	FDA Parameters Short interval Long interval		30 12			minutes			
						hours			

7.4.6 Other VUE Settings

When VUE is launched, it will automatically open the last database used if "Automatically load previous database when VUE is launched" is selected in the *Other* tab. This is a useful tool if you are using the same database or if your database is not large. Simply unselect the option and VUE will not load a database when it is launched.

The *Other* tab also gives you the option to restore the VUE settings to their default values and to restore all the hidden dialogs (see receiver manual for more information on hiding dialogs).

Dptions Display Offload Export Devices Release	s FDA Tool Other
Database Image: Constraint of the second s	when VUE is launched
Defaults Pestore Default Options	This can be unselected if you are working with a large database (saves time when opening VUE).
Restore All Hidden Dialogs	Hidden dialogs are restored automatically whenever one week elapses between uses of VUE.
	OK Cancel

8. APPENDICES

8.1 Contact Information

Innovasea

20 Angus Morton Drive Bedford, Nova Scotia Canada B4B 0L9

Phone: +1.902.450.1700 Email: support.team@innovasea.com

support.fishtracking.innovasea.com
innovasea.com/fish-tracking/

